INTEGRATED QUALITY AND ENVIRONMENTAL MANAGEMENT FRAMEWORK FOR SMALL AND MEDIUM FOOD PROCESSING ENTERPRISES IN PAKISTAN

SYED ALI RAZA SHAH

UNIVERSITI TEKNOLOGI MALAYSIA

INTEGRATED QUALITY AND ENVIRONMENTAL MANAGEMENT FRAMEWORK FOR SMALL AND MEDIUM FOOD PROCESSING ENTERPRISES IN PAKISTAN

SYED ALI RAZA SHAH

A thesis submitted in fulfilment of the requirements for the award of the degree of Doctor of Philosophy

Razak Faculty of Technology and Informatics Universiti Teknologi Malaysia

DEDICATION

This thesis is dedicated to my (late) father, who taught me that the best kind of knowledge to have is that which is learned for its own sake. It is also dedicated to my mother, who taught me that even the largest task can be accomplished if it is done one step at a time.

ACKNOWLEDGEMENT

Alhamdulillah, it has been an overwhelming intellectual and emotional journey. Almighty Allah has been the most gracious to have blessed me in innumerable ways to complete this mighty challenge to complete my Ph.D. No doubt, I was blessed with very competent and able supervisors Associate Professor Dr. Khairur Rijal Jamaludin and Dr. Hayati Habibah Abdul Talib. I am very much thankful for their tremendous support and guidance as well as precious suggestions to uplift the quality of my thesis. Their willingness to share his/her knowledge and expertise not only aided me in achieving my goal but also broadened my research horizons. Truly, without their exalted guidance and supervision it would not have been possible for me to complete my thesis. I feel very privileged and blessed to have the golden opportunity to work with such great personalities.

My special gratitude goes to Dean Razak Faculty of Technology and Informatics, Prof. Dr. Ir Sha'ri Mohammad Yusof for his valuable support and guidance especially in improving my publication skills. I am also very much thankful to all the staff of Razak Faculty of Technology and Informatics for their guidance and support whenever I needed, special thank goes to Dr. Jihad Mohammad for helping me in PLS analysis. I would like to extend my appreciation and thanks to Higher Education Commission of Pakistan and Balochistan University of Engineering and Technology Khuzdar, Pakistan for provision of scholarship for PhD studies. I also want to say thanks to the officials of Small and Medium Enterprise Development Authority and Pakistan Chamber of Commerce and Industry for their help in data collection. To all my friends, I thank you for providing me with the moral comfort that helped my go on whenever I felt exhausted, few important names Mohammad Arif, Syed Azhar Karam Shah, and Mohammad Shakeel.

Most importantly I want to thank my mother, brothers, and sisters, whose unstoppable prayers, made it possible for me to complete this never ending and emotionally exhausting journey. Finally I would like to say special thanks to my wife, as she scarified a lot with her serious illness. She always motivated me, giving me strength to complete this task and enriching me morally so I could forget everything and concentrate on the completion of my doctoral degree. I dedicate all these years of hard work to my wife and my great kids Abeera Shah, Syed Waleed Raza, and Faria Raza.

ABSTRACT

Food is an essential component for our daily dietary needs. The demand for food is increasing as with the increasing population. Undoubtedly, the significance of food industry to the countries' growth is well recognized. However, as reported in the literature, majority of food industry players are Small and Medium Enterprises (SMEs), where they are confronted with limitations and constraints to succeed in the business growth due to limited resources. Nowadays, food industry has been reported to consume abundant resources of water and energy, and pollute the environment through water waste, solid wastes, and air pollution. Therefore, further study on these related issues should be explored particularly in the developing countries. Thus, the research was decided to be conducted among food processing SMEs in Pakistan. Pakistan is a developing country where the implementation of quality and environmental management practices is far behind. The awareness and implementation of environmental regulations are also reported weak, especially by the Pakistan's SME. Currently, the country is facing hindrances in global food trade due to the lack of environmental concerns. To date, there are no empirical studies on quality and environmental management in food processing SMEs of Pakistan, and the extent of integrated quality and environmental management practices is still a grey area. Moreover, the important role of organization culture is also less studied in developing countries such as Pakistan. To fill the research gaps, this study aims to identify the Critical Success Factors (CSFs) of Integrated Quality and Environmental Management (IQEM) and examine their impact on the performance of food processing SMEs in Pakistan. The moderating role of organizational culture on the relationship between IQEM practices and the organizational performance is also examined. Seven constructs are used to measure IOEM whereas the organizational performance is measured by operational and environmental performances. A questionnaire was developed and distributed to food processing SMEs operating in Punjab, Pakistan, which is 65.27 % of the total SMEs are in Pakistan. The unit of analysis is SME company and the respondents of the questionnaire are those who possess knowledge about the subject matter. Purposive with snowball sampling techniques were used for data collection. A total of 302 food processing SME owners or managers responded to the survey but only 288 are useable for further analysis. IBM-SPSS version-23 and SmartPLS-3 were used for data analysis and hypothesis testing. The results show that CSFs of IQEM have significant impact to the performance of food processing SMEs in Pakistan. Moreover, the organizational culture moderates the relationship between IQEM and the organizational performance. This study has succeeded to fill the research gap identified, especially on the CSFs of IQEM for food processing SMEs in the developing countries such as Pakistan. It is believed that by practicing CSFs of IOEM, food SMEs can improve their performance and increase their competitiveness. Furthermore, organizational culture is also revealed as a key factor that is significant on organizational performance. The research findings will help the SMEs to increase their implementation awareness on IQEM practices to gain success in business. Indirectly, this study will be able to help food processing SMEs especially in Pakistan to improve their holistic organizational performance. It will also be able to motivate and expose a role of organizational culture to be imposed by SME food industry especially in Pakistan in order to gain competitive advantage.

ABSTRAK

Makanan adalah komponen penting bagi keperluan diet harian. Permintaan untuk makanan semakin meningkat seiring dengan peningkatan populasi penduduk. Tidak dinafikan, kepentingan industri makanan terhadap pertumbuhan negara telah diiktiraf dengan baik. Walau bagaimanapun, seperti yang dilaporkan dalam literatur, majoriti pemain industri makanan adalah Perusahaan Kecil dan Sederhana (SMEs), di mana mereka menghadapi pelbagai batasan dan kekangan untuk berjaya dalam perniagaan disebabkan oleh sumbersumber yang terhad. Pada masa kini, industri makanan juga telah dilaporkan menggunakan banyak sumber air dan tenaga, dan mencemarkan alam sekitar melalui sisa air, sisa pepejal, dan pencemaran udara. Oleh itu, kajian lanjut tentang isu-isu berkaitan perlu diterokai terutamanya di negara-negara sedang membangun. Dengan yang demikian, kajian ini diputuskan untuk dijalankan dalam kalangan SMEs pemprosesan makanan di Pakistan. Pakistan adalah sebuah negara yang sedang membangun yakni pelaksanaan pengurusan kualiti dan alam sekitar adalah jauh kebelakang. Kesedaran dan pelaksanaan peraturan alam sekitar juga dilaporkan lemah, terutama oleh SMEs di Pakistan. Pada masa ini, negara menghadapi halangan dalam perdagangan makanan global disebabkan kurang memberi perhatian terhadap alam sekitar. Sehingga kini, tiada kajian empirikal tentang pengurusan kualiti dan alam sekitar SMEs pemprosesan makanan di Pakistan, dan sejauh mana integrasi pengurusan kualiti alam sekitar diamalkan masih kabur. Selain itu, kepentingan peranan budaya organisasi juga kurang dikaji di negara-negara membangun seperti Pakistan. Untuk merapatkan jurang kajian, kajian ini bertujuan untuk mengenal pasti Faktor Kejayaan Kritikal (CSFs) Intergrasi Pengurusan Kualiti dan Alam Sekitar (IQEM) dan mengkaji impaknya terhadap prestasi SMEs pemprosesan makanan di Pakistan. Peranan penyederhanaan budaya organisasi ke atas hubungan antara amalan IQEM dengan prestasi organisasi turut dikaji. Tujuh konstruk digunakan untuk mengukur IQEM manakala, prestasi organisasi diukur oleh prestasi operasi dan persekitaran. Soal selidik telah dibangunkan dan diedarkan kepada SMEs pemprosesan makanan yang beroperasi di Punjab, Pakistan, iaitu 65.27% daripada jumlah SMEs di Pakistan. Unit analisis bagi soal selidik ini adalah syarikat SMEs dan responden yang berpengetahuan tentang perkara ini. Teknik pensampelan bola salji digunakan untuk pengumpulan data. Sejumlah 302 pemilik atau pengurus SMEs pemprosesan makanan telah menjawab kaji selidik tersebut namum hanya 288 sahaja yang boleh digunakan untuk tujuan analisis selanjutnya. IBM-SPSS versi-23 dan SmartPLS-3 digunakan untuk menganalisis data dan menguji hipotesis. Keputusan menunjukkan semua CSF IQEM mempunyai impak yang signifikan terhadap prestasi SMEs pemprosesan makanan di Pakistan. Selain itu, budaya organisasi menyederhanakan hubungan antara IOEM degan prestasi organisasi. Kajian ini telah berjaya mengisi jurang penyelidikan yang telah dikenalpasti, terutamanya pada elemen CSF bagi IQEM untuk SMEs pemprosesan makanan di negara-negara membangun seperti Pakistan. Adalah dipercayai bahawa dengan mengamalkan CSFs bagi IQEM, SMEs makanan dapat meningkatkan prestasi dan daya saing. Tambahan pula, budaya organisasi juga telah dibuktikan sebagai faktor utama yang signifikan ke atas prestasi organisasi. Dapatan kajian ini akan dapat membantu SMEs meningkatkan kesedaran terhadap pelaksanaan amalan IQEM untuk mencapai kejayaan dalam perniagaan. Secara tidak langsung, kajian ini juga dapat membantu SMEs pemprosesan makanan terutamanya di Pakistan meningkatkan prestasi organisasi secara holistik. La juga dapat memotivasikan dan mendedahkan peranan budaya organisasi yang akan dilaksanakan oleh SMEs makanan terutamanya di Pakistan untuk mencapai kelebihan bersaingan.

TABLE OF CONTENTS

TITLE

DECLARATION	iii
DEDICATION	iv
ACKNOWLEDGEMENT	v
ABSTRACT	vi
ABSTRAK	vii
TABLE OF CONTENTS	viii
LIST OF TABLES	XV
LIST OF FIGURES	xviii
LIST OF ABBREVIATIONS	XX
LIST OF SYMBOLS	xxii
LIST OF APPENDICES	xxiii

CHAPTER 1	INTRODUCTION	1
1.1	Background of Study	1
1.2	Problem Statement	8
1.3	Research Questions	12
1.4	Research Objectives	12
1.5	Research Scope	13
1.6	Significance of the Research	13
1.7	Definition of Terms	14
1.8	Outline of the Thesis	15
1.9	Summary	16
CHAPTER 2	LITERATURE REVIEW	17
2.1	Introduction	17
2.2	Food Industry	17
	2.2.1 Overview of Pakistan's Food industry	18
	2.2.2 Environmental Issues in Food Industry	20

2.3	Small	and Medium-Sized Enterprises	22
	2.3.1	Definition of Small and Medium-Sized Enterprises in Pakistan	24
	2.3.2	Provincial Location of Small and Medium- Sized Enterprises in Pakistan	26
	2.3.3	Small and Medium Enterprise Development Authority (SMEDA)	27
	2.3.4	Food Processing Small and Medium-sized Enterprises	28
	2.3.5	Small and Medium-Sized Enterprises Issues in Pakistan	29
2.4	Resea	rch Gap	30
2.5	Explo	ratory Study	35
2.6	Qualit	ty Management	39
	2.6.1	The Quality Guru's and their Contributions	40
	2.6.2	Quality Awards	44
	2.6.3	Critical Success Factors (CSFs)	45
	2.6.4	Quality Management in Food Industry	46
	2.6.5	Quality Management Critical Success Factors	47
2.7	Envir	onmental Management	49
	2.7.1	Environmental Management in Food Industry	51
	2.7.2	Environmental Management Critical Success Factors	51
2.8	Integr	rated Management Systems (IMS)	53
	2.8.1	Integrated Quality and Environmental Management (IQEM)	54
	2.8.2	Integrated Quality Environmental Management Critical Success Factors	58
	2.8.3	Comparison of Identified Critical Success Factors in Quality Awards	60
	2.8.4	Proposed Description of Critical Success Factors of Integrated Quality Environmental Management for Small and Medium-Sized Enterprises in Pakistan	62
		2.9.4.1 Londorship	05 C A
		2.0.4.1 Leadership	64
		2.8.4.2 Employee Management	65

			2.8.4.3	Strategic Planning	66
			2.8.4.4	Information Management	66
			2.8.4.5	Process Management	67
			2.8.4.6	Supplier Management	68
			2.8.4.7	Customer Focus	69
		2.8.5	Underpi	nning Theories	70
	2.9	Organ	izational	Culture	71
	2.10	Perfor	mance		73
		2.10.1	Operatio	onal Performance	74
		2.10.2	Environ	mental performance	75
	2.11	Resea	rch Frame	ework	77
		2.11.1	IQEM a	nd Performance	78
		2.11.2	Moderat	ing Role of Organizational Culture	79
	2.12	Summ	nary		82
СНАРТЕ	R 3	MET	HODOLO	DGY	83
	3.1	Introd	uction		83
	3.2	Resea	rch Parad	igm	83
	3.3	Resea	rch Desig	n	84
	3.4	Data (Collection		87
		3.4.1	Survey		88
		3.4.2	Populati	on and Sampling	88
		3.4.3	Samplin	g Techniques	91
	3.5	Instru	ment Dev	elopment	95
		3.5.1	Construe	ct Measurement for IQEM	96
			3.5.1.1	Leadership (LS)	96
			3.5.1.2	Employee Management (EM)	97
			3.5.1.3	Strategic Planning (SP)	97
			3.5.1.4	Information Management (IM)	98
			3.5.1.5	Process Management (PM)	99
			3.5.1.6	Supplier Management (SP)	99
			3.5.1.7	Customer Focus	100

	3.5.2	Construct Measurement of Organizational Culture (OC)	101
	3.5.3	Construct Measurement of Performance	101
		3.5.3.1 Operational Performance (OP)	102
		3.5.3.2 Environmental Performance (EP)	102
	3.5.4	Pilot Study	103
	3.5.5	Reliability and Validity	105
		3.5.5.1 Content Validity	105
		3.5.5.2 Construct Validity	106
3.6	Data A	Analysis	106
	I.	Rule of Thumb for Selecting CB-SEM or PLS-SEM	108
	II.	Partial Least Square-Structural Equation Modelling	109
3.7	Summ	nary	112
CHAPTER 4	DATA	A ANALYSIS	113
4.1	Introd	uction	113
4.2	Analy	sis of Respondents Background	113
4.3	Respo	ondents Demographics	114
	4.3.1	Education Background of Respondents	114
	4.3.2	Working Experience of Respondents in Food Processing SMEs	115
	4.3.3	Current Working Department or Unit	116
	4.3.4	Type of Food Product Manufacturing	116
	4.3.5	Year of Establishment	117
	4.3.6	Number of Permanent Employees	118
	4.3.7	Nature of SME Ownership	118
	4.3.8	SME Location	119
	4.3.9	Export Involvement	120
4.4	Data S	Screening and Preliminary Analysis	121
	4.4.1	Missing Data Analysis	122
	4.4.2	Outliers Detection	123
	4.4.3	Common Method Bias	124

	4.4.4	Normali	ty Assessment	125
4.5	PLS S	EM Appr	oach	130
	4.5.1	Research	n Model Analysis	130
	4.5.2	Assessm Model-F	ent of Measurement Model (Outer First Order)	132
		4.5.2.1	Internal Consistency Reliability	134
		4.5.2.2	Convergent Validity	134
		4.5.2.3	Discriminant Validity	136
	4.5.3	Assessm	ent of Second-Order Constructs	139
		4.5.3.1	Assessment of Second-Order Formative (IQEM) Construct	142
		4.5.3.2	Multicollinearity	143
		4.5.3.3	Outer Weights	144
		4.5.3.4	Outer loadings	144
	4.5.4	Assessm	ent of Structural Model (Inner Model)	146
		4.5.4.1	Path Coefficient (β) and t-values	147
		4.5.4.2	Explanatory Power of the Model	148
		4.5.4.3	Effect Size (f^2)	149
		4.5.4.4	Predictive Relevance (Q ²)	150
	4.5.5	Moderat	ion Test	151
		4.5.5.1	Moderating effect of Organizational Culture on the relationship between IQEM and Operational Performance	152
		4.5.5.2	Moderating effect of Organizational Culture on the relationship between IQEM and Environmental Performance	153
	456	Goodnee	as of Fit	155
	457	Summar	v Overall Hypothesis	156
4 6	Summ	narv	y overall hypothesis	150
1.0	2 withit	J		107
CHAPTER 5	RESU	JLTS DIS	SCUSSION	159
5.1	Introd	uction		159
5.2	Result	ts Finding	s based on Research Objectives	159

5.3	Resear Succes Enviro proces	rch Objective 1: To determine the Critical ss Factors (CSFs) of Integrated Quality and onmental Management (IQEM) among food using SMEs in Pakistan.	161
	5.3.1	Importance of CSFs in Quality Management Practices	162
	5.3.2	Importance of CSFs in Environmental Management Practices	163
	5.3.3	Importance of CSFs in Integrated Quality and Environmental Management (IQEM)	164
5.4	Reseau integra (IQEN Pakist	rch Objective 2: To examine the relationship of ated quality and environmental management 1) and performance in food processing SMEs in	165
	5 A 1	an. Urmathagia 1: Thara is a positiva relationship	105
	3.4.1	between IQEM and operational performance in food processing SMEs in Pakistan	166
	5.4.2	Hypothesis 2: There is a positive relationship between IQEM and environmental performance in food processing SMEs in Pakistan	168
5.5	Resear of org betwee manag proces	rch Objective 3: To analyse the moderating role ganizational culture (OC) on the relationship en integrated quality and environmental gement (IQEM) and performance in food using SMEs in Pakistan.	170
	5.5.1	Hypothesis 3: Organizational culture (OC) moderates the relationship between IQEM and operational performance (OP) in food processing SMEs in Pakistan	170
	5.5.2	Hypothesis 4: Organizational culture moderates the relationship between IQEM and environmental performance in food processing SMEs in Pakistan.	172
5.6	Resear impler proces culture	rch Objective 4: To propose a framework for the mentation of IQEM in Pakistan's food using SMEs by considering local organizational	175
57	Summ	arv	178
5.7	2 unin	j	1,0
CHAPTER 6	CON	CLUSION AND RECOMMENDATIONS	179
6.1	Introd	uction	179

6.2	Research Objectives Fulfillment	179
6.3	Contributions of the Study	181
	6.3.1 Contributions to the Knowledge	181
	6.3.2 Contributions to the Practices	183
	6.3.2.1 To the Country	183
	6.3.2.2 To SMEs	184
6.4	Limitations and Challenges of the Study	185
6.5	Recommendations	187
6.6	Summary	187
REFERENCES		189
LIST OF PUBLI	CATIONS	246

LIST OF TABLES

TABLE NO.	TITLE	PAGE
Table 2.1	Water Usage in Various Food and Beverage Processing	21
Table 2.2	Importance of Small and Medium-sized Enterprises to the National Economies	22
Table 2.3	Definition of Small and Medium-Sized Enterprises based on Number of Employees	23
Table 2.4	Definitions of Small and Medium-sized Enterprises based on Number of Employees in Some Asian Countries	24
Table 2.5	Definitions of Small and Medium-Sized Enterprises in Pakistan	25
Table 2.6	Previous Studies related to Food SMEs	30
Table 2.7	Previous Studies Related to QM, EM and TQM Practices in Pakistan	32
Table 2.8	Summary of Small and Medium-Sized Enterprises involved in Exploratory Study	37
Table 2.9	Quality Management Gurus and their Contributions	40
Table 2.10	Quality Improvement Stages	41
Table 2.11	Critical Success Factors of Quality Management Proposed by Previous Studies	48
Table 2.12	Critical Success Factors of Environmental Management Proposed by Previous Studies	52
Table 2.13	Significance of Quality Management on Environmental Management Practices by Previous Studies	55
Table 2.14	Comparison of Critical Success Factors in Quality and Environmental Management Studies	59
Table 2.15	Comparison of Identified Critical Success Factors of IQEM in Corresponding Quality Awards	60
Table 2.16	Comparison of Names of Critical Success Factors in QM and EM Practices and Proposed Names of Critical Success Factors for IQEM	61
Table 2.17	Description of Proposed Critical Success Factors of IQEM for Food Processing SMEs in Pakistan	63

Table 2.18	Operational Performances Measurement in Different Studies	75
Table 2.19	Environmental Performances Measurement in Different Studies	76
Table 3.1	Sample Size Recommendations in Previous Studies	90
Table 3.2	Items to Measure Leadership Construct	96
Table 3.3	Items to Measure Employee Management Construct	97
Table 3.4	Items to Measure Strategic Planning Construct	98
Table 3.5	Items to Measure Information Management Construct	98
Table 3.6	Items to Measure Process Management Construct	99
Table 3.7	Items to Measure Supplier Management Construct	99
Table 3.8	Items to Measure Customer Focus Construct	100
Table 3.9	Items to Measure Organizational Culture Construct	101
Table 3.10	Items to Measure Operational Performance Construct	102
Table 3.11	Items to Measure Environmental Performance Construct	103
Table 3.12	Cronbach Alpha Values (Pilot and Final Data)	104
Table 3.13	Assessment Criteria for Reflective and Formative constructs	111
Table 4.1	Summary of Final Data Sheet	114
Table 4.2	Respondents Education Background	114
Table 4.3	Working Experience in Food Processing SMEs	115
Table 4.4	Current Working Department or Unit in this SME	116
Table 4.5	Type of Manufacturing/Processing	117
Table 4.6	Year of Establishment	117
Table 4.7	Number of Permanent Employees	118
Table 4.8	Nature of SME Ownership	118
Table 4.9	SMEs Location	119
Table 4.10	Export Involvement	120
Table 4.11	Data Analysis Sections and Procedures	121
Table 4.12	Item-wise Missing Values	122
Table 4.13	Results of Harmon Single Factor Test	125

Table 4.14	Results of Skewness and Kurtosis	126
Table 4.15	Construct Types	132
Table 4.16	Internal Consistency Reliability and Convergent Validity Results	135
Table 4.17	Fornell and Larcker Criterion	136
Table 4.18	Cross Loadings	137
Table 4.19	Heterotrait-Monotrait Ratio (HTMT)	139
Table 4.20	Results of Variance Inflated Factor, Significance of Outer Weights and Significance of Outer Loadings of Formative Construct IQEM	145
Table 4.21	Hypothesis Relationship among Constructs	147
Table 4.22	Results for coefficient of Determination (R ²)	149
Table 4.23	Results for Effect Size (f^2)	149
Table 4.24	Predictive Relevance Q ²	150
Table 4.25	Hypothesis Testing for Moderation with Operational Performance	152
Table 4.26	Hypothesis Testing for Moderation	154
Table 4.27	Model Fit Value	156
Table 4.28	Results Summary of Hypothesis Testing	156
Table 6.1	Summary of Research Objectives Fulfilment and Achievements	180

LIST OF FIGURES

FIGURE NO	. TITLE	PAGE
Figure 2.1	Food Processing and Related Sectors with Products Source:SMEDA (2010b)	20
Figure 2.2	Provincial Location of Small and Medium-Sized Enterprises in Pakistan Source: SMEDA (2009)	27
Figure 2.3	Conceptual Framework	77
Figure 3.1	Research Design and Flow Chart	86
Figure 3.2	Assessment of Structural Model Source: Hair et al. (2017)	110
Figure 3.3	Reflective and Formative Constructs Source: Vinzi <i>et al.</i> (2010)	111
Figure 4.1	Regression Standardized Residual of Independent and Dependent Variable	128
Figure 4.2	Normal P-p plot of Regression Standard Residual	129
Figure 4.3	Regression Standardized Residual of Independent and Dependent Variable	129
Figure 4.4	Normal P-p plot of Regression Standard Residual	130
Figure 4.5	Reflective-Formative Higher Order Model	131
Figure 4.6	Measurement Model	133
Figure 4.7	Reflective-Reflective Model (Type-1) Source: Hair <i>et al.</i> (2014)	140
Figure 4.8	Reflective-Formative Model (Type-II) Source: Hair <i>et al.</i> (2014)	140
Figure 4.9	Formative-Reflective Model (Type-III) Source: Hair <i>et al.</i> (2014)	141
Figure 4.10	Formative-Formative Model (Type-IV) Source: Hair <i>et al.</i> (2014)	141
Figure 4.11	Second-Order Formative Model with Latent Variable Score	143
Figure 4.12	Significance of Outer Weights	145
Figure 4.13	Structural Model Assessment Procedure	146
Figure 4.14	Structural Model	148

Figure 4.15	Blindfolding				150
Figure 4.16	Moderation Model Organizational Culture	IQEM-OP	Moderated	by	153
Figure 4.17	Moderation Model Organizational Culture	IQEM-EP	Moderated	by	154
Figure 4.18	Interaction Plot				155

LIST OF ABBREVIATIONS

BC	-	Before Christ
BOD	-	Biochemical Oxygen Demand
CBM-	-	Covariance based Structural Equation Modelling
SEM		
CI		Confidence interval
CMB		Common Method Bias
CR	-	Composite Reliability
COD	-	Chemical Oxygen Demand
CSFs	-	Critical Success Factors
DP	-	Deming Prize
EFQM	-	European Foundation for Quality Management
EM	-	Environmental Management
EU	-	European Union
FBS	-	Federal Beuro of Statistics
FSM		Food Safety Management
GDP		Gross Domestic Product
НАССР		Hazard Analysis Critical Control Point
HRM		Human Resource Management
HTMT		Heterotrait-Monotrait Ratio
IPMA		Importance performance Map Analysis
IQEM		Integrated Quality Environmental Management
ISO		International Organization for Standardization
MBNQA		Malcolm Baldridge National Quality Award
MDPR		Ministry of Planning, Development and Reforms
MOCC		Ministry of Climate Change
PBS		Pakistan Bureau of Statics
PDCA		Plan, Do, Check, Act
PLS-		Partial Least Squares-Structural Equation Modelling
SEM		
PNAC		Pakistan National Accreditation Council

PSQCA	Pakistan Standards and Quality Control Authority
QM	Quality Management
RBV	Resource-based View
SBP	State Bank of Pakistan
SCM	Supply Chain Management
SECP	Securities and Exchange Commission of Pakistan
SEM	Structural Equation Modelling
SHT	Stake Holder Theory
SMEs	Small and Medium-Sized Enterprises
SMEDA	Small and Medium-Sized Enterprises Development
	Authority
SPSS	Statistical Package for the Social Sciences
SRMR	Standardized Mean Square Residual
TQM	Total Quality Management
UNIDO	United Nations Industrial Development Organization
UTM	Universiti Teknologi Malaysia
WB	World Bank
WTO	World Trade Organization

LIST OF SYMBOLS

- α Cronbach alpha
- β Path coefficient
- f^2 Effect size
- Q² Predictive relevance
- R² Coefficient of determination

LIST OF APPENDICES

APPENDIX	TITLE	PAGE
Appendix A	Letter to SMEDA	229
Appendix B	Letter to PSQCA	230
Appendix C	Questionnaire	231
Appendix D	Data Collection Letter	237
Appendix E	Pilot Study Letter	238
Appendix F	Chamber of Commerce Letter	239
Appendix G	Summary of Experts Opinion	240
Appendix H	Z Scores	244

CHAPTER 1

INTRODUCTION

1.1 Background of Study

Food is an important component for our body and used in daily life for its dietary needs. Safe food keeps us healthy and prevents foodborne diseases. Food processing is a technology which adds value to increase the shelf-life of the products (Fellows, 2009). The food processing industry mainly converts agro-materials into the final products by using several process design techniques (Jonkman *et al.*, 2017). The history of food processing and preservation techniques starts from around 3000-1500BC where the Egyptians first developed techniques such as sun drying for poultry and fish, ovens for baking, fermentation and cereal grinding. Later, some pastoral societies preserved food for use in times of famine. With the passage of time, food technologies developed in many places such as China, Japan, Europe and India, where they invented different kinds of food processing technologies according to their needs and food preferences (Fellows, 2009).

The demand for processed food is increasing globally due to its unique characteristics, such as easy cooking, which requires less time and consumes less energy, and the provision of products with a longer shelf-life and a range of different varieties and flavours (Boyue and Arcand, 2010; Chukwu, 2009). Augustin *et al.* (2016), stated that the world population is growing fast and is estimated to be 9 billion in 2050; besides the other challenges, the production of food is believed to be a big challenge as the growing population will need more food, water and resources. This is supported by Oldfield, White, and Holden (2016), who note that, due to increasing demand for food, meeting the sustainability needs of the population is a big challenge.

Food systems are at the heart of the 2030 Agenda for Sustainable Development (UN, 2015), a global commitment to eradicate poverty and hunger while ensuring reduction of environmental and socio-economic impacts (Sala *et al.*, 2016). The developments in resource availability, consumer trends and legislation require the food industry to regularly evaluate its mode of operations in order to remain competitive in global markets (Jonkman *et al.*, 2017). Food processing is considered as one of the solutions that reduces food wastage and provides longer shelf-life to the products.

Pakistan is an agrarian country and its food industry is considered the country's largest with an export potential of 13% of total exports (Trading Economics, 2018). The food and beverage processing industry accounts for approximately 27% of total production, 16% of the total manufacturing industry manpower is employed in this sector (SMEDA, 2010b). Pakistan's Small and Medium-Sized Enterprises (SMEs) make up 90% of the country's businesses and play an important role in socio-economic development, (Khan, Awang, and Zulkifli, 2013). Food and beverage SMEs make up 20.09% of total manufacturing and come second after the textile sector (Ahmad, Naz, and Majid, 2018; SMEDA, 2009).

Pakistan's SMEs have great export potential, however, currently, SMEs are facing different kinds of problem such as a shortage of skilled labour, lack of infrastructure, lack of education and training, non-compliance with standards, export barriers, fierce international competition and an energy crisis. These problems and issues are hindering the performance of SMEs (Dar, Ahmed, and Raziq, 2017; Khan and Khalique, 2014; Khattak, Arslan, and Umair, 2011). According to Dar *et al.* (2017) and Rehman (2016), little research and development work has been conducted in the SME sector. According to the best of the researcher's knowledge, no noteworthy studies on food processing SMEs in Pakistan have been conducted. Therefore, food processing SMEs are a grey area.

Beside the advantages of processed food, this industry is considered as one of the major sources of environmental pollution through solid waste, air pollution, noise pollution and waste water (Pipatprapa, Huang, and Huang, 2016). Numerous authors highlighted the environmental problems/issues in food processing such as inefficiencies in food production by producing more waste, consumption of abundant resources of water and energy, food losses in different supply chains, food packing and the transportation of food. These inefficient production practices are imposing negative impacts on the natural environment (Abeele *et al.*, 2017; Alsaffar, 2016; Goot *et al.*, 2016). Moreover, Adeyeye (2017) stated that the world is facing food security challenges due to inefficient production practices; therefore, countries need to be more focused on sustainable production practices since a country's economic growth and sustainability is highly dependent on food security. Similarly, Notarnicola *et al.* (2016) stated that there is a dire need to fulfil the fundamental human need for nutrition; hence, ensuring food security requires a revised research agenda.

Moreover, globalization, intense competition and ever-changing customer demands have dramatically changed the business environment during the last few decades (Ebrahimi and Sadeghi, 2013). In this regard, fulfilling the customer's demand for higher product quality is becoming challenging; therefore, organizations are seeking to improve their products' quality *as per* customers' requirements. Thus, adoption of quality improvement initiatives is increasing day-by-day. Similarly, the sustainability issues and environmental pollution have influenced organizations to reduce their environmental burden and focus more on sustainable production (Mezinska and Strode, 2015). According to Ali and Suleiman (2016), in the food sector, consumers are now demanding more stringent quality control, food safety and healthy food along with environmental compliance.

According to Singh, Jain, and Sharma (2015), a growing number of companies are now recognizing the importance of sustaining the natural environment and trying to reduce environmental impacts from their operations; for that purpose, companies are developing new environmental strategies and programs in order to get higher environmental benefits since the industrial sector is considered as one of the major sources of pollution. Therefore, industries and their stakeholders are focusing more on their production practices in order to make the operations more sustainable (Alayón, Säfsten, and Johansson, 2017; de Sousa Jabbour *et al.*, 2018; Shafiq,

Lasrado, and Hafeez, 2017). In addition, Saad (2016) stated that organizations are seeking to optimize their performance in order to deliver high-quality products, reduce waste, enhance the recycling of materials, minimise operating costs, protect the environment, conserve resources and promote economic feasibility. In this regard triple bottom-line concepts such as people, planet and profit are considered vital for organizations that provide social, environmental and financial benefits (Alayón *et al.*, 2017).

In the early 2000s, the adoption of management systems for quality, the environment and safety became an important activity for many organizations since these management systems are considered as a symbol of success that provides the opportunity to control the risks, costs, environmental impacts, non-conformities and customer satisfaction whilst undertaking a continuous improvement approach (Moumen and El Aoufir, 2017). Moreover, quality management focuses on process improvement, reduction of wastage and continuous improvement. It provides superior value to the customers and improves the efficiency of the processes whereas environmental management focuses on reducing pollution, air emissions, solid waste and hazardous waste (Alayón *et al.*, 2017; Bernardo *et al.*, 2015; Miles and Covin, 2000).

The environmental performance is determined as how company improves its performance by reducing the solid wastes, reducing the water wastage, decrease the consumption of harmful material, reduce the consumption of water and energy by using efficient equipment. Moreover; quality and environmental management systems were acknowledged as important tools to put the sustainable production principles into practice (Alayón *et al.*, 2017). Implementing quality and environmental management practices within an organization helps create a sustainable business (Antony, 2015). In addition, the protection of the natural environmental practices are becoming key instruments and have received much critical attention from researchers and practitioners (Hamdoun, Chiappetta Jabbour, and Ben Othman, 2018). Moreover, sustainable manufacturing practices have been defined mostly from an environmental perspective, aiming at minimizing the impacts of manufacturing operations on the environment while optimizing the production efficiency of the company (Alayón *et al.*, 2017; Nordin, Ashari, and Rajemi, 2014).

Furthermore, the concepts of integration of management systems are becoming a very popular research topic they enhance an organization's efficiency and performance and provide benefit from the synergies existing among them (Simon, Karapetrovic, and Casadesús, 2012). An integrated management system (IMS) is a single management system that delivers the processes of the business through mutually supportive structured management functions configured around the wider needs of the organization (Griffith and Bhutto, 2009). Previous studies revealed that IMSs provide cost savings, avoid duplication of tasks, improve corporate image, improve performance and provide higher customer satisfaction (Molina-Azorín, Tarí, *et al.*, 2009; Rebelo, Santos, and Silva, 2016; Tarí and Molina-Azorín, 2010; Vílchez and Darnall, 2016).

Quality Management is considered to provide suitable support for the integration of sustainability considerations in the area of product development. Moreover, QM can be integrated with other management systems to achieve sustainability, such as an environmental management system (Siva *et al.*, 2016), occupational health and safety management systems (Santos *et al.*, 2013), corporate social responsibility (Khurshid, Amin, and Ismail, 2018) and supply chain management (Bastas and Liyanage, 2019). According to Oskarsson and Malmborg (2005), the most commonly integrated management systems are quality and environmental management systems that are adjusted to function together. By integrating QM and EM, an all-embracing, effective, single, integrated, quality environmental management (IQEM) system is created (Bernardo *et al.*, 2018; Molina-Azorín, Tarí, *et al.*, 2009; Tarí and Molina-Azorín, 2010).

According to Karapetrovic and Willborn (1998), an integrated quality and environmental management system improves technology development and transfer, improves operational performance, improves internal management methods, enhances the confidence of customers, provides a positive market image, reduces costs and is more efficient for re-engineering. The literature revealed that integrated quality environmental management practices provide sustainability is operations, increase performance, improve corporate image and provide higher customer satisfaction. Therefore, this study is focused on integrated quality environmental management (IQEM) in food processing SMEs and identifies the critical success factors (CSFs) of IQEM, in order to improve the performance of food processing SMEs since Pakistan's food processing SMEs are lacking in terms of quality and environmental performance. According to Ram and Corkindale (2014), CSFs help to identify the key areas which require constant and careful attention for achieving organizational goals thereby ensuring the organizations' competitive performance.

It is important to mention that for the implementation of quality, environmental or integrated management initiatives, organizational culture plays a pivotal role. Culture has been identified as one of the key determinants of the success that provides sustainable economic and business connections among countries and industries (Nguyen and Aoyama, 2015). According to Uzkurt *et al.* (2013), organizations seek to improve their performance and, in this regard, organizational culture has been recognized as one of the important drivers to improve a firm's performance. Furthermore, Wong-MingJi *et al.* (2014) mentioned that cross-cultural boundaries have an impact on the development of high-quality, productive, international businesses.

Discussing the role of organizational culture on quality initiatives, Jabnoun and Khafaji (2005) mentioned that culture is a key element in determining success or failure of quality initiatives, even in the adoption of QM programs which also vary from country to country and between organizations. According to Prajogo and McDermott (2011), organizational culture provides different competitive performance priorities by identifying the specific cultural dimensions. Therefore, organizations might choose to position how to compete in particular cultural characteristics. As mentioned by Prajogo and McDermott (2005), several organizations failed to achieve the expected goals of the implementation of quality management as a result of ignoring cultural factors. This is supported by Aziz, Maria, and Rahayu (2017) and Arshad, Halipah, and Omar (2018), where they noted that many companies fail to implement quality management because they do not recognize that the implementation of the procedure may require a fundamental change of direction for the values and culture of that company.

The literature review shows that organizational culture plays a key role in achieving organizational success. Therefore, any change in organizational culture can lead to the success or failure of the system. If employees of the company are willing to accept the change, especially for quality and environmental initiatives, the company will definitely achieve its targets and become competitive in the market, otherwise there is a chance of failure of the system. Rezaei et al. (2018), stated that organizational culture is considered as a strategic source for 21st century success in both business and the economy, it is also one of the fundamentals for sustainable competitive advantage. Thus, organizational culture plays a fundamental role in achieving sustainable environmental management practices and the implementation of sustainable practices needs proper embedding within an organizational culture (Bakhsh Magsi et al., 2018). According to Tayeb (1994), 93% of studies concluded that culture is an important factor and cultural values, traditions, behaviours and attitudes play a vital role in shaping an organization. Therefore, the role of organizational culture cannot be neglected, especially for the implementation of IQEM practices where the role of organizational culture is crucial.

Moreover, this study is underpinned by two theories namely: Resource-based View (RBV), and stake holder theory (SHT). RBV is the most widely accepted theory that focuses on internal resources and capabilities of an organization; as internal resources are considered as the main source of an organization's success (Akio, 2005). Moreover, stakeholders are defined as, any group or individual who can affect or is affected by the achievement of the organization's objectives (Hadi, Abdullah, and Sajilan, 2015). Since, both RBV and SHT theories fulfil the objectives of this study, and provide support to improve the performance, provides higher customer satisfaction, and positive company image. Likewise, both QM and EM practices have significant impact on performance and provides higher customer satisfaction and positive company image.

1.2 Problem Statement

There are several quality and environment problems and issues in food processing SMEs that are affecting their performance in Pakistan. According to the Government of Punjab (2015), constraints such as access to international markets, lower levels of standardization of quality assurance as well as corruption are hindering industrial performance. The poor quality control mechanisms and applied standards create difficulties in exporting products, whilst firms seeking certification are using labs outside Pakistan in order to meet international requirements for their export orders. Furthermore, highlighting the quality problems, the State Bank of Pakistan (2017) has reported that maintaining quality in food products is a major challenge for the food processing industry since food industries needs to adopt quality standards throughout their supply chains in order to achieve a competitive environment and customer satisfaction.

In addition, Kureshi *et al.* (2010) stated that the SME sector is reported to be less involved in quality management practices. Moreover, most of the SMEs in the country have minimal quality systems and their product quality is either not being ensured or being ensured with some informal practices. According to Irfan *et al.* (2014), Pakistan is considered to be a late adopter of total quality management (TQM) practices, only adopting TQM in 2010. Discussing the performance of SMEs in Pakistan, Bhutta, Rana, and Asad (2007) stated that their performance is not up to the mark, indeed it is showing a decreasing trend. This is supported by Khalique *et al.* (2015) where they stated that it is an alarming situation for the Pakistani economy because poor performance of SMEs is associated with poor performance for industrial output.

Furthermore, environmental issues have gained more attention in the last few decades due to the increasing number of global warming and climate change predictions. The media, civil society, Non-Governmental Organizations (NGOs), Environmental Protection Agencies (EPA) and governmental departments are pressurizing organizations to address their environmental issues and improve their environmental performance. For these reasons, the global community is more aware

about environmental issues and prefers the products that fulfil both quality and environmental compliance (Li *et al.*, 2017; Oliveira, Serra, and Salgado, 2010; Quader, Kamal, and Hassan, 2016; Sharma, Chandna, and Bhardwaj, 2017).

It is pertinent to mention that losses through environmental degradation in Pakistan are estimated to be around Rs.365 billion PKR per year (Sánchez-Triana *et al.*, 2014), which is almost equal to 6% of GDP (Ortolano *et al.*, 2014). The implementation of environmental regulations is considered weak (Nadeem and Hameed, 2008; Sánchez-Triana *et al.*, 2014). This is supported by Ortolano *et al.* (2014) where they stated that there is a lack of environmental awareness and that most of the firms do not know that the country has environmental regulations that they supposed to meet. Due to unavailability of authentic data related to sectorial contributions towards environment pollution, it is difficult to mention the exact contribution of food processing SMEs, however, food and beverage processing SMEs come second after the textile sector within manufacturing SMEs (Ahmad *et al.*, 2018).

The food and beverage processing industry is considered as one of the major sources of environmental pollution and the increasing demand for processed food also exerts stress on energy and water resources (Zhang and Vesselinov, 2017). Food industries produce environmental pollution through food processing losses, food waste, packaging, low energy efficiency, transportation of food, water consumption and producing more waste (Alsaffar, 2016). Furthermore, water, energy and raw materials are the main resources used in the food processing industry. The waste water from food and beverage processing industries contains lot of organic compounds, high concentrations of Biochemical Oxygen Demand (BOD), Chemical Oxygen Demand (COD), Suspended Solids (SS), total Nitrogen (N), total Phosphorous (P) and with some heavy metals also present, all of which are harmful for health and the natural environment (Abeele *et al.*, 2017). This is supported by Goot et al. (2016), where they stated that current food production practices have a large impact on the environment through inefficiencies in food production that produce more waste and pollution. According to the national round-table on sustainable, consumption and production (SCP) (Switch Asia (2016), Pakistan is behind in resource efficiency as compared to other Asian countries in terms of poor drinking water quality, an energy crisis and also due to the inefficient production and consumption practices.

It is important to mention that the weak implementation and non-compliance with, QM and EM practices are not only hindering the performance of food processing SMEs in Pakistan; they are also considered as serious obstacles to exports, especially by the European Union (EU), the USA and Japan, which are particularly concerned about environmental compliance. Pakistan faces significant non-tariff barriers to trade related to poor performance in the area of quality, hygiene and Sanitary and Phytosanitary (SPS) measures. The country has faced a number of bans in the past; among others, fish and seafood exports were banned by the European Union in 2007, due to unhygienic conditions at Karachi fish harbour (USAID, 2007). It is evident that some countries rejected the consignments due to non-compliance with standards (Trade Related Technical Assistance, 2012). The literature reveals that the food processing industry is one of the major sources that pollute the environment. Therefore, the country needs to transform from a traditional developing country market with low quality and low-priced products to one that is high-quality, safe and healthy, with environmental compliance for food products in order to fulfil the sustainability requirements.

Moreover; organizational culture is considered an important factor towards organizational performance. It is important to mention that many of the QM failure were due to the lack of understanding of the culture of that organization. Thus, organizational culture is also a significant factor towards the organizational performance and needs to further investigate to the causal relationship on the IQEM practices among food processing SMEs in Pakistan. It is pertinent to mention that organizational culture is less studied in developing countries such as Pakistan, and food processing SMEs are the grey area. There is a dire need to further investigate this important factor since this is to reveal that organizational culture is an external factor in contributing IQEM as success factor. This study has selected organizational culture as a moderator variable since exploratory results confirmed that in Pakistan's food processing SMEs the organizational culture is weak in term of quality management, environmental management, and performance and also the employees were found less educated. There were lack of awareness about environmental regulation and the implementations of environmental regulations were found weak. Therefore, it is hypothesized that there is a third variable that influencing the performance of SMEs. According to Baron and Kenny (1986), when the relationship between a predictor and a criterion variable is found unexpectedly weak or inconsistent, a moderating variable should be introduced. Since, IQEM is formed by the CSFs of QM and EM and the previous results of both QM and EM practices shows inconsistency in results therefore organizational culture is introduced as moderating variable, as confirmed by exploratory study and previous literature.

Thus, above mentioned issues and problem related to food processing SMEs need to be solved and given the priority, as the food sector of the country is important sector and have a great export potential. In conclusion to the above discussion, both operational and environmental performances of food processing SMEs in Pakistan are not up to the mark. Since, previous literature revealed that both quality and environmental management practices have a significant impact on both operational and environmental performance. Therefore; this study is focused on identifying the critical success factors for IQEM, in which quality and environmental aspects are covered within a single system and had an impact on organizational performance. This study examines IQEM impact on the operational and environmental performance of food processing SMEs. This study also examines the moderation role of organizational culture on the relationship between IQEM, and operational and environmental performance. By addressing quality, environmental and export issues, this study provides some valuable insights for Pakistan's food manufacturers that on the one hand address the quality and environmental issues whilst on the other hand encouraging the practise of CSFs for IQEM so that food players can enhance their trade with the rest of the world where, currently, the country is facing hindrances due to quality and environmental concerns. Similarly, by understanding the relationship of IQEM and organizational culture food players can improve the organizational performance through a good organizational culture by the involvement of all employees.

1.3 Research Questions

Research questions allow researchers to interpret the research problem into the need for investigation (Bilal, 2014; Ruane, 2005). The following research questions are guiding this study:

- 1. What are the Critical Success Factors (CSFs) of Integrated Quality Environmental Management (IQEM) practices among food processing SMEs in Pakistan?
- 2. How do the critical success factors of IQEM contribute to the performance of food processing SMEs in Pakistan?
- 3. Does Organizational Culture (OC) moderate the relationship between IQEM and the performance of food processing SMEs in Pakistan?

1.4 Research Objectives

The research objectives demonstrate the rationale of the study and describe the standards and ways to accomplish the research work (Bilal, 2014). The following are the objectives for this study:

- 1. To determine the critical success factors of IQEM among food processing SMEs in Pakistan.
- 2. To examine the relationship of IQEM and performance in the food processing SMEs in Pakistan.
- 3. To analyze the moderating role of organizational culture on the relationship between IQEM and performance in food processing SMEs in Pakistan.
- 4. To propose a framework for the implementation of IQEM in Pakistan's food processing SMEs by considering local organizational culture.

1.5 Research Scope

This study focuses on food processing SMEs, since the demand for processed food is increasing globally with the increasing population (Augustin *et al.*, 2016). The processed food provides products with a longer shelf-life that are available in different varieties and flavours (Fellows, 2009). Pakistan, being a developing country, is focusing on industrial development, particularly on the SME sector, and the food and beverage industry is the largest industry within manufacturing with a share of 20.09% (Ahmad *et al.*, 2018).

This study has collected data from food processing SMEs located in Punjab province, Pakistan. The reason for choosing Punjab province is due to the large number of SMEs, around 65.27% of the total SMEs of the country are located in Punjab province (Khan *et al.*, 2013; SMEDA, 2009). Moreover, 60% of food industries are also located in this province (Punjab Skill Development Fund, 2015). This study uses a quantitative methodology and the data is collected through a survey questionnaire. Purposive with snowball sampling techniques are used to select the respondents. Each SME received one questionnaire and the respondents of the study are owners, managers, chief executive officers (CEOs) or a person who have knowledge about QM, EM and the IQEM practices and performance of the company.

1.6 Significance of the Research

Like any research the aim of this study is to make a meaningful contribution at theoretical and empirical levels. The findings contribute to the body of knowledge in the field of integrated quality environmental management (IQEM), by identifying the critical success factors of IQEM practices, especially in the food processing SMEs in Pakistan. This study examines the impact of IQEM on the operational and environmental performance. Also, the moderating role of organizational culture on the relationship between IQEM and operational and environmental performance is examined. This study proposes a framework for IQEM implementation by considering local organizational culture in order to solve problems concerning quality and the environment in the food processing SMEs and hence to reach a better level of quality and environmental management and to achieve a high quality in production.

This study is very important due to its significance in addressing the quality and environmental issues. Furthermore, by understanding the relationship of the CSFs of IQEM and organizational culture, food players would be able to implement effective IQEM and improve their operational and environmental performance. Moreover, the findings help food manufacturers to enhance their trade with the rest of the world, especially the European Union (EU), where the country is facing obstacles due to quality and environmental concerns. This study helps the country to fulfil the World Trade Organization (WTO) prerequisites for conducting international trade. Finally, the findings of the study enable the nation to achieve Pakistan's Vision 2025 targets, which require the country to reduce the population subject to food insecurity and increase the country's exports (MPDR, 2016).

1.7 Definition of Terms

a) Quality Management (QM) and Total Quality Management (TQM)

QM and TQM are customer-focused management philosophies and strategies that seek continuous improvement in businesses and processes by using different tool and techniques. The concepts of QM and TQM are almost the same and the ultimate goal of both is same such as continuous improvement and customer focus; thus this study uses the term QM instead of TQM.

b) Environmental Management (EM)

EM provides a set of management tools and principles that can help integrate environmental concerns into daily business practices, EM enhances the environmental performance by reducing the usage of natural resources.

c) Integrated Quality Environmental Management (IQEM)
IQEM is an integrated management approach that integrate quality and environmental management in order to enhance the overall operating and environmental efficiency of the organizations

d) Critical Success Factors (CSF)

CSFs are the analytical tool to evaluate the characteristics of an industry in which it competes. CSFs are the essential prerequisite to control the implementation process in order to increase the chances of success.

e) Operational Performance (OP)

Operational performance is defined as the capability of manufacturing companies to optimize the production process in order to improve product quality and assure on time delivery.

f) Environmental Performance (EP)

Environmental performance is defined as the ability of manufacturing companies to reduce their environmental impact by decreasing the consumption of hazardous and toxic materials, air emission, waste water and solid waste.

g) Organizational Culture (OC)

Organizational culture has been recognized as one of the important drivers to better the firm performance. Cultural norms, beliefs, behaviours, attitudes and customs have an important role that helps in forming a good organizational culture.

h) Partial Least Square-Structure Equation Modelling (PLS-SEM)

PLS-SEM is second-generation multivariate statistical analysis technique. It is used to test the causal relationships, it maximise the explained variance of the dependent variable

1.8 Outline of the Thesis

This thesis is organized into six chapters. The first chapter has highlighted the background of the research. It has also outlined the problem statement, research

questions, research objectives, scope and the significance of the research. Chapter Two reviews the literature on the food industry and its importance, the environmental issues in food processing, small and medium-sized enterprises, food processing SMEs and identifies the research gap. It also includes exploratory study, quality management, environmental management and integrated quality environmental management, critical success factors, organizational culture, performance and its measures and is followed by the research framework and hypotheses.

Chapter Three illustrates the methods and procedures that will be used in this study. It gives a detailed description of the research process including the research paradigm, research design and flow and the research methodologies. This chapter also explains data collection, the survey population and sampling, the sampling techniques, unit of analysis, instrument development, the constructs and their measurement, the pilot study, reliability and validity and data analysis. Chapter Four presents the data analysis including primary data analysis for screening and cleaning of data through IBM-SPSS followed by hypothesis-testing through Partial Least Squares-Structure Equation Modeling (PLS-SEM).

Chapter Five provides an overall discussion of the research findings in the light of the results of Chapter Four. Finally, Chapter Six concludes with the contributions of the study, to the body of knowledge and practice. This chapter also presents the limitations of the study and provides recommendations for future research.

1.9 Summary

This chapter has provided the background of the study, its importance and the increasing demand for food, the history of food processing, the environmental issues in food processing, small and medium-sized enterprises and food processing SMEs. It also presented the problem statement, research questions, research objectives, research scope and the significance of the research in detail. It also presented an outline of the thesis.

REFERENCES

Abad, J., Dalmau, I., & Vilajosana, J. (2014). Taxonomic proposal for integration levels of management systems based on empirical evidence and derived corporate benefits. *Journal of Cleaner Production*, 78, 164-173.

Abeele, L. V. d., Smets, T., Derden, A., Huybrechts, D., & Nevens, F. (2017). Feasibility study for the food processing industry in Flanders (Belgium) to become water neutral by 2030. *Journal of Cleaner Production*, *141*, 1376-1390.

- Aboulnaga, I. A. (1998). Integrating quality and environmental management as competitive business strategy for 21st century. *Environmental Management and Health*, 9(2), 65-71.
- Adeyeye, S. A. (2017). The role of food processing and appropriate storage technologies in ensuring food security and food availability in Africa. *Nutrition & Food Science*, 47(1).
- Agus, A., & Hassan, Z. f. (2011). Enhancing Production Performance and Customer Performance Through Total Quality Management (TQM): Strategies For Competitive Advantage. *Procedia - Social and Behavioral Sciences, 24*, 1650-1662.
- Ahire, S. L., Golhar, D. Y., & Waller, M. A. (1996). Development and Validation of TQM Implementation Constructs. *Decision Sciences*, 27(1), 23-56.
- Ahmad, B., Naz, F., & Majid, J. (2018). Growth Prospects of SMEs in CPEC Perspective. Global Management Journal for Academic & Corporate Studies, 8(1), 101-111.
- Ahmad, M. S. (2012). Impact of Organizational Culture on Performance Management Practices in Pakistan. *Business Intellegence Journal*, 5(1), 50-55.
- Ahmed, A., Khuwaja, F. M., Brohi, N. A., & Othman, I. b. L. (2018). Organizational Factors and Organizational Performance: A Resource-Based view and Social Exchange Theory Viewpoint. *International Journal of Academic Research in Business and Social Sciences*, 8(3).

- Aibinu, A. A., & Al-Lawati, A. M. (2010). Using PLS-SEM technique to model construction organizations' willingness to participate in e-bidding. *Automation in Construction*, 19(6), 714-724.
- Akio, T. (2005). The Critical Assessment of the Resource-Based View of Strategic Management: The Source of Heterogeneity of the Firm. *Ritsumeikan International Affairs*, 3, 125-150.
- Al-Swidi, A. K., & Mahmood, R. (2012). Total quality management, entrepreneurial orientation and organizational performance: The role of organizational culture. *African Journal of Business Management*, 6(13), 4717-4727.
- Al-Tuwaijri, S. A., Christensen, T. E., & Hughes, K. E. (2004). The relations among environmental disclosure, environmental performance, and economic performance: a simultaneous equations approach. *Accounting, Organizations* and Society, 29(5), 447-471.
- Alayón, C., Säfsten, K., & Johansson, G. (2017). Conceptual sustainable production principles in practice: Do they reflect what companies do? *Journal of Cleaner Production*, 141, 693-701.
- Alghamdi, F. (2018). Total Quality Management and Organizational Performance: A Possible Role of Organizational Culture. *International Journal of Business Administration*, 9(4), 186-200.
- Ali, M. H., & Suleiman, N. (2016). Sustainable food production: Insights of Malaysian halal small and medium sized enterprises. *International Journal of Production Economics*, 181, 303-314.
- Alonso-Almeida, M. d. M., Rodríguez-Antón, J. M., & Rubio-Andrada, L. (2012). Reasons for implementing certified quality systems and impact on performance: an analysis of the hotel industry. *The Service Industries Journal*, 32(6), 919-936.
- Alsaffar, A. A. (2016). Sustainable diets: The interaction between food industry, nutrition, health and the environment. *Food Science and Technology International*, 22(2), 102-111.
- Amaratunga, D., Baldry, D., Sarshar, M., & Newton, R. (2002). Quantitative and qualitative research in the built environment: application of "mixed" research approach. *Work Study*, 51(1), 17-31.
- Amin, M., Aldakhil, A. M., Wu, C., Rezaei, S., & Cobanoglu, C. (2017). The structural relationship between TQM, employee satisfaction and hotel

performance. International Journal of Contemporary Hospitality Management, 29(4), 1256-1278.

- Amjad, R., Ghani, E., Din, M. u., & Mahmood, T. (2012). Export Barriers in Pakistan: Results of a Firm-Level Survey. *The Lahore Journal of Economics*, 17, 103-134.
- Anderson, J. C., & Gerbing, D. W. (1988). Structural Equation Modeling in Practice: A Review and Recommended Two-Step Approach. *Psychological Bulletin*, 103(3), 411-423.
- Anderson, J. C., Rungtusanatham, M., Schroeder, R. G., & Devaraj, S. (1995). A Path Analytic Model of a Theory of Quality Management Underlying the Deming Management Method: Preliminary Empirical Findings. *Decision Sciences*, 26(5), 637-658.
- Antony, J. (2015). The ten commandments of quality: a performance perspective. International Journal of Productivity and Performance Management, 64(5), 723-735.
- Aquilani, B., Silvestri, C., & Ruggieri, A. (2016). Sustainability, TQM and Value Co-Creation Processes: The Role of Critical Success Factors. *Sustainability*, 8(10), 995.
- Aquilani, B., Silvestri, C., Ruggieri, A., & Gatti, C. (2017). A systematic literature review on total quality management critical success factors and the identification of new avenues of research *The TQM Journal*, *29*(1), 184-213.
- Arda, O. A., Bayraktar, E., & Tatoglu, E. (2019). How do integrated quality and environmental management practices affect firm performance? Mediating roles of quality performance and environmental proactivity. *Business Strategy* and the Environment, 28(1), 64-78.
- Arshad, K. N. M., Halipah, A. H., & Omar, R. C. (2018). The Effect of Organizational Culture towards Total Quality Management (TQM) Implementation in Malaysia Public Organization. *International Academic Journal of Accounting and Financial Management*, 5(3), 1-11.
- Asim, M., Zaman, S. u., & Zarif, T. (2013). Implementation of Total Quality Management in Construction Industry: A Pakistan Perspective. *Journal of Management and Social Sciences*, 9(1), 24-39.

- Augustin, M. A., Riley, M., Stockmann, R., Bennett, L., Kahl, A., Lockett, T., . . . Cobiac, L. (2016). Role of food processing in food and nutrition security. *Trends in Food Science & Technology*, 56, 115-125.
- Avolio, B. J., Yammarino, F. J., & Bass, B. M. (1991). Identifying Common Methods Variance With Data Collected From A Single Source: An Unresolved Sticky Issue. *Journal of Management*, 17(3), 571-587.
- Awan, M. U., Raouf, A., Ahmed, N., & Sparks, L. (2009). Total quality management in developing countries: A case of pharmaceutical wholesale distribution in Pakistan. *International Journal of Pharmaceutical and Healthcare Marketing*, 3(4), 363-380.
- Aziz, R. A., Maria, D., & Rahayu, S. (2017). Implementation of Total Quality Management (TQM), Organizational Culture and Performance: Evidence from Indonesian Micro, Small and Medium Enterprises (MSMEs). Proceeding of Endinamosis 2017.
- Aziz, S., Mahmood, M., & Bano, S. (2018). Total Quality Management: A Frame Work for Higher Education Institution. *Journal of Research in Social Sciences*, 6(1), 121-141.
- Bahri, S., Hamzah, D., & Yusuf, R. M. (2012). Implementation of Total Quality Management and Its Effect on Organizational Performance of Manufacturing Industries Through Organizational Culture in South Sulawesi, Indonesia. *IOSR Journal of Business and Management*, 5, 10-24.
- Baig, S. A., Zia-ur-Rehman, M., Amjad, F., Ali, I., Hashim, M., & Yousaf, S. (2017). Impact of Quality Management Practiceson Performance: Moderating Role of Innovation Culture. *Journal of Managerial Sciences*, 11(3), 391-409.
- Bakhsh Magsi, H., Ong, T., Ho, J., & Sheikh Hassan, A. (2018). Organizational Culture and Environmental Performance. *Sustainability*, 10(8), 2690.
- Barney, J. (1991). Firm Resources and Sustained Competitive Advantage. Journal of Management, 17(1), 99-120.
- Baron, R. M., & Kenny, D. A. (1986). The Moderator-Mediator Variable Distinction in Social Psychological Research: Conceptual, Strategic, and Statistical Considerations. *Journal of Personality and Social Psychology*, 51(6), 1173-1182.
- Başaran, B. (2016). The effect of ISO quality management system standards on industrial property rights in Turkey. *World Patent Information, 45*, 33-46.

- Bastas, A., & Liyanage, K. (2019). Integrated quality and supply chain management business diagnostics for organizational sustainability improvement. *Sustainable Production and Consumption*, 17, 11-30.
- Basu, R., & Bhola, P. (2016). Impact of quality management practices on performance stimulating growth: Empirical evidence from Indian IT enabled service SMEs. *International Journal of Quality & Reliability Management*, 33(8), 1179-1201.
- Beauregard, T. A., Basile, K. A., & Thompson, C. A. (2018). Organizational culture in the context of national culture. In R. Johnson, W. Shen, & K. M. Shockley (Eds.), The Cambridge handbook of the global work-family interface (pp. 555-569). Cambridge: Cambridge University Press.
- Becker, J.-M., Klein, K., & Wetzels, M. (2012). Hierarchical Latent Variable Models in PLS-SEM: Guidelines for Using Reflective-Formative Type Models. *Long Range Planning*, 45(5), 359-394.
- Benner, M. J., & Veloso, F. M. (2008). ISO 9000 practices and financial performance: A technology coherence perspective. *Journal of Operations Management*, 26(5), 611-629.
- Bentler, P. M. (1990). Comparative fit indexes in structural models. *Psychological Bulletin*, 107(2), 238-246.
- Bentler, P. M., & Chou, C.-P. (1987). Practical Issues in Structural Modeling. SOCIOLOGICAL METHODS & RESEARCH, 16(1), 78-117.
- Bernardo, M., Casadesus, M., Karapetrovic, S., & Heras, I. (2012). Integration of standardized management systems: does the implementation order matter? *International Journal of Operations & Production Management*, 32(3), 291-307.
- Bernardo, M., Gotzamani, K., Vouzas, F., & Casadesus, M. (2018). A qualitative study on integrated management systems in a non-leading country in certifications. *Total Quality Management & Business Excellence*, 29(3-4), 453-480.
- Bernardo, M., Simon, A., Tarí, J. J., & Molina-Azorín, J. F. (2015). Benefits of management systems integration: a literature review. *Journal of Cleaner Production*, 94, 260-267.

- Berry, M. A., & Rondinelli, D. A. (1998). Proactive corporate environmental management: A new industrial revolution. Academy of Management Perspectives, 12(2), 38-50.
- Bhutta, M. K. S., Rana, A. I., & Asad, U. (2007). SCM practices and the health of the SMEs in Pakistan. Supply Chain Management: An International Journal, 12(6), 412-422.
- Bhutta, M. K. S., Rana, A. I., & Asad, U. (2008). Owner characteristics and health of SMEs in Pakistan. *Journal of Small Business and Enterprise Development*, 15(1), 130-149.
- Bilal, A. R. (2014). "Financial Practices and Performance of Small and Medium Enterprises in Pakistan" PhD Thesis, Universiti Teknologi Malaysia.
- Boeck, E. D., Jacxsens, L., Mortier, A. V., & Vlerick, P. (2018). Quantitative study of food safety climate in Belgian food processing companies in view of their organizational characteristics. *Food Control*, 88, 15-27.
- Boolen, K. A. (1989). *Structural Equations with Latent Variables*. New York: John Wileyand Sons, Inc.
- Boomsma, A. (1982). The robustness of LISREL against small sample sizes in factor analysis models. *Systems under indirect observation : Causality, structure, prediction, 1*, 149-173.
- Boyue, J. I., & Arcand, Y. (2010). *Green Technologies in Production and Food Processing*: Springer New York Dordrecht Heidelberg London.
- Bremmers, H., Omta, O., Kemp, R., & Haverkamp, D.-J. (2007). Do stakeholder groups influence environmental management system development in the Dutch agri-food sector? *Business Strategy and the Environment, 16*(3), 214-231.
- Bryman, A., & Bell, E. (2015). *Business research methods*: Oxford University Press, USA.
- Bureauo of Statistics Punjab. (2017). Punjab Development Statistics. Retrieved 25-15-2017, from <u>http://bos.gop.pk/system/files/PDS%202017_0.pdf</u>
- Burns, K. E. A., Duffett, M., Kho, M. E., Meade, M. O., Adhikari, N. K. J., Sinuff, T., & Cook, D. J. (2008). A guide for the design and conduct of selfadministered surveys of clinicians. *Canadian Medical Association Journal*, 179(3), 245-252.

- Burrows, P. (1992). TQM reality check: it works, but it's not cheap or easy. *Electronic Business, 18*, 8-22.
- Calvo-Mora, A., Domínguez-Cc, M., & Criado, F. (2018). Assessment and improvement of organisational social impact through the EFQM Excellence Model. *Total Quality Management & Business Excellence*, 29(11-12), 1259-1278.
- Calza, F., Cannavale, C., & Tutore, I. (2016). The important effects of national culture on the environmental proactivity of firms. *Journal of Management Development*, 35(8), 1011-1030.
- Campos, L. M. S., de Melo Heizen, D. A., Verdinelli, M. A., & Cauchick Miguel, P.
 A. (2015). Environmental performance indicators: a study on ISO 14001 certified companies. *Journal of Cleaner Production*, 99, 286-296.
- Carpinetti, L. C. R., Buosi, T., & Gerólamo, M. C. (2003). Quality management and improvement: A framework and a business-process reference model. *Business Process Management Journal*, 9(4), 543-554.
- Casadesús, M., Karapetrovic, S., & Heras, I. (2011). Synergies in standardized management systems: some empirical evidence. *The TQM Journal*, *23*(1), 73-86.
- Cassells, S., Lewis, K., & Findlater, A. (2011). SMEs and ISO 14001 adoption: A New Zealand perspective. *Small Enterprise Research*, 18(1), 19-32.
- Champney, H., & Marshall, H. (1939). Optimal refinement of the rating scale. Journal of Applied Psychology, 23(3), 323-331.
- Chin, K.-S., Chiu, S., & Tummala, V. M. R. (1999). An evaluation of success factors using the AHP to implement ISO 14001-based EMS. *International Journal of Quality & Reliability Management*, 16(4), 341-362.
- Chin, W. W. (1998a). Commentry: Issues and Opinion on Structural Equation Modeling. *MIS Quarterly*, 22(1), 7-16.
- Chin, W. W. (1998b). The Partial Least Squares Approach to Structural Equation Modeling. *Modern methods for business research*, 295(2), 295-336.
- Chin, W. W. (2010). "How to write up and report PLS analyses" in EspositoVinzi, V., Chin, W.W., Henseler, J. and Wang, H. (Eds), Handbook of Partial Least Squares: Concepts, Methods and Applications. *Springer, Heidelberg*, pp. 655-690.

- Chin, W. W., & Newsted, P. R. (1999). Structural Equation Modeling Analysis with Small Samples Using Partial Least Square. *Statistical strategies for small sample research*, 1(1), 307-341.
- Choudhary, A. I., Akhtar, S. A., & Zaheer, A. (2013). Impact of transformational and servant leadership on organizational performance: A comparative analysis. *Journal of Business Ethics*, 116(2), 433-440.
- Chuang, F.-M., Morgan, R. E., & Robson, M. J. (2012). Clan culture, strategic orientation and new product performance in Chinese marketing ventures: an exploration of main and moderating effects. *Journal of Strategic Marketing*, 20(3), 267-286.
- Chukwu, O. (2009). Impacts of food processing industry on some environmental health and safety factors. *Caspian Journal of Environmental Sciences*, 7(1), 37-44.
- Cohen, J. (1988). Statistical power analysis for the behavioral sciences. Hilsdale. NJ: Lawrence Earlbaum Associates.
- Collis, J., & Hussey, R. (2003). Business Research , Palgrave, London.
- Cooper, D. R., & Schindler, P. S. (2006). Marketing research (p. 261). New York: McGraw-Hill/Irwin.
- Cordeiro, J. J., & Sarkis, J. (1997). Environmental proactivism and firm performance: evidence from security analyst earnings forecasts. *Business Strategy and the Environment, 6*(2), 104-114.
- Corredor, P., & Goñi, S. (2011). TQM and performance: Is the relationship so obvious? *Journal of Business Research*, 64(8), 830-838.
- Costin, H. (1994). Total Quality Management, Dryden, USA.
- Creswell, J. W. (2009). *Research design: Qualitative, quantitative, and mixed methods research*: Los Angeles: SAGE.
- Creswell, J. W. (2013). Research design: Qualitative, quantitative, and mixed methods approaches: Sage Publications.
- Creswell, J. W., & Creswell, J. D. (2017). *Research design: Qualitative, quantitative, and mix method approaches:* Sage Publication.
- Cuerva, M. C., Triguero-Cano, Á., & Córcoles, D. (2014). Drivers of green and nongreen innovation: empirical evidence in Low-Tech SMEs. *Journal of Cleaner Production, 68*, 104-113.

- Curkovic, S. (2003). Environmentally Responsible Manufacturing: The development and validation of a measurement model. *European Journal of Operational Research*, *146*(1), 130-155.
- Danyen, S., & Callychurn, D. S. (2015). Total quality management success factors and their relationships with performance measures in the food industry: a Mauritian case study. *Int. J. Productivity and Quality Management, 16*(3), 249-266.
- Dar, M. S., Ahmed, S., & Raziq, A. (2017). Small and Medium- Size enerprises in Pakistan: Definition and Critical Issues. *Pakistan Business Review*, 46-70.
- Das, A., Paul, H., & Swierczek, F. W. (2008). Developing and validating total quality management (TQM) constructs in the context of Thailand's manufacturing industry. *Benchmarking: An International Journal*, 15(1), 52-72.
- Davis, D. L. (1998). Business Research for Decision Making With Infotrac. Brooks/Cole Publishing Company.
- Dawson, J. F. (2014). Moderation in Management Research: What, Why, When, and How. *Journal of Business and Psychology*, 29(1), 1-19.
- De Smedt, D., Clays, E., Doyle, F., Kotseva, K., Prugger, C., Pająk, A., . . . De Bacquer, D. (2013). Validity and reliability of three commonly used quality of life measures in a large European population of coronary heart disease patients. *International Journal of Cardiology*, *167*(5), 2294-2299.
- de Sousa Jabbour, A. B. L., Jabbour, C. J. C., Foropon, C., & Godinho Filho, M. (2018). When titans meet – Can industry 4.0 revolutionise the environmentally-sustainable manufacturing wave? The role of critical success factors. *Technological Forecasting and Social Change*, 132, 18-25.
- Demirbag, M., Tatoglu, E., Tekinkus, M., & Zaim, S. (2006). An analysis of the relationship between TQM implementation and organizational performance: Evidence from Turkish SMEs. *Journal of Manufacturing Technology Management*, 17(6), 829-847.
- Denison, D. R., & Mishra, A. K. (1995). Toward a Theory of Organizational Culture and Effectiveness. Organization Science, 6(2), 204-223.
- Diamantopoulos, A., Riefler, P., & Roth, K. P. (2008). Advancing formative measurement models. *Journal of Business Research*, *61*(12), 1203-1218.

- Diana, G. C., Jabbour, C. J. C., de Sousa Jabbour, A. B. L., & Kannan, D. (2017). Putting environmental technologies into the mainstream: Adoption of environmental technologies by medium-sized manufacturing firms in Brazil. *Journal of Cleaner Production*, 142, 4011-4018.
- Ding, L., Velicer, W. F., & Harlow, L. L. (1995). Effects of estimation methods, number of indicators per factor, and improper solutions on structural equation modeling fit indices. *Structural Equation Modeling: A Multidisciplinary Journal*, 2(2), 119-143.
- Djekic, I. (2015). Environmental Impact of Meat Industry Current Status and Future Perspectives. *Procedia Food Science*, *5*, 61-64.
- Dooyoung, S., Kalinowski, J. G., & El-Enein, G. (1998). Critical implementation issues in total quality management. SAM Advanced Management Journal, 63(1), 10-14.
- Dora, M., Kumar, M., & Gellynck, X. (2015). Determinants and barriers to lean implementation in food-processing SMEs – a multiple case analysis. *Production Planning & Control, 27*(1), 1-23.
- Douglas, A., & Glen, D. (2000). Integrated management systems in small and medium enterprises. *Total Quality Management*, 11(4-6), 686-690.
- Duarte, P., & Amaro, S. (2018). Methods for modelling reflective-formative second order constructs in PLS: An application to online travel shopping. *Journal of Hospitality and Tourism Technology*, 9(3), 295-313.
- Dubey, R., Gunasekaran, A., Helo, P., Papadopoulos, T., Childe, S. J., & Sahay, B. S. (2017). Explaining the impact of reconfigurable manufacturing systems on environmental performance: The role of top management and organizational culture. *Journal of Cleaner Production*, 141(2017), 56-66.
- Ebrahimi, M., & Sadeghi, M. (2013). Quality management and performance: An annotated review. *International Journal of Production Research*, *51*(18), 5625-5643.
- EFQM. (2018). European Foundation for Quality Management. Retrieved 10-01-2019, from <u>http://www.efqm.org/</u>
- Famiyeh, S., Adaku, E., Amoako-Gyampah, K., Asante-Darko, D., & Amoatey, C. T. (2018). Environmental management practices, operational competitiveness and environmental performance: Empirical evidence from a developing country. *Journal of Manufacturing Technology Management, 29*(3), 588-607.

- Faul, F., Erdfelder, E., Buchner, A., & Lang, A.-G. (2009). Statistical power analyses using G* Power 3.1: Tests for correlation and regression analyses. *Behavior Research Methods*, 41(4), 1149-1160.
- Fellows, P. J. (2009). *Food processing technology principles and practice* (Third ed.): Woodhead Publishing Limited.
- Fening, F. A., Pesakovic, G., & Amaria, P. (2008). Relationship between quality management practices and the performance of small and medium size enterprises (SMEs) in Ghana. *International Journal of Quality & Reliability Management, 25*(7), 694-708.
- Fernando, Y., Wah, W. X., & Shaharudin, M. S. (2016). Does a firm's innovation category matter in practising eco-innovation? Evidence from the lens of Malaysia companies practicing green technology. *Journal of Manufacturing Technology Management*, 27(2), 208-233.
- Figge, F. (2005). Value-based environmental management. From environmental shareholder value to environmental option value. *Corporate Social Responsibility and Environmental Management, 12*(1), 19-30.
- Flower Jr, F. J. (2013). Survey research methods: Sage Publications.
- Flynn, B. B., Schroeder, R. G., & Sakakilbara, S. (1995). The Impact of Quality Management Practices on Performance and Competitive Advantage. *Decision Sciences*, 26(5), 659-691.
- Fornell, C., & Bookstein, F. L. (1982). Two Structural Equation Models: LISREL and PLS Applied to Consumer Exit-Voice Theory. *Journal of Marketing Research*, 19(4), 440-452.
- Fornell, C., & Larcker, D. F. (1981). Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. *Journal of Marketing Research*, 18(1), 39-50.
- Fotopoulos, C. V., & Psomas, E. L. (2010). The structural relationships between TQM factors and organizational performance. *The TQM Journal, 22*(5), 539-552.
- Freund, Y. P. (1988). Critical success factors. *Planning Review*, 16(4), 20-23.
- García-Bernal, J., & Ramírez-Alesón, M. (2015). Why and How TQM Leads to Performance Improvements. *Quality Management Journal*, 22(3), 23-37.
- Garland, R. (1991). The mid-point on a rating scale: Is it desirable. *Marketing bulletin*, 2(1), 66-70.

- Garner, W. R. (1960). Rating scales, discriminability, and information transmission. *Psychological Review*, 67(6), 343-350.
- Garvin, D. A. (1988). Management Quality. The free Press, New York.
- Gilley, K. M., Worrell, D. L., Davidson, W. N., & El–Jelly, A. (2000). Corporate Environmental Initiatives and Anticipated Firm Performance: The Differential Effects of Process-Driven Versus Product-Driven Greening Initiatives. *Journal of Management*, 26(6), 1199-1216.
- Gimenez-Espin, J. A., Jiménez-Jiménez, D., & Martínez-Costa, M. (2013). Organizational culture for total quality management. *Total Quality Management & Business Excellence*, 24(5-6), 678-692.
- Goetsch, d. L., & Davis, S. (2013). *Quality Management for Organizational Excellence* (Seventh Edition ed.): London, PEARSON.
- Goot, A. J. v. d., Pelgrom, P. J. M., Berghout, J. A. M., Geerts, M. E. J., Jankowiak, L., Hardt, N. A., . . . Boom, R. M. (2016). Concepts for further sustainable production of foods. *Journal of Food Engineering*, 168, 42-51.
- Government of Punjab. (2015). Punjab Growth Strategy 2018: Accelerating Economic Growth and Improving Social Outcomes. Retrieved 27-6-2017, from https://www.theigc.org/wp-content/uploads/2015/04/Punjab-Growth-Strategy-2018-Full-report.pdf
- Graafland, J., & Smid, H. (2017). Reconsidering the relevance of social license pressure and government regulation for environmental performance of European SMEs. *Journal of Cleaner Production*, 141, 967-977.
- Graham, S., & McAdam, R. (2016). The effects of pollution prevention on performance. *International Journal of Operations & Production Management*, 36(10), 1333-1358.
- Graham, S., & Potter, A. (2015). Environmental operations management and its links with proactivity and performance: A study of the UK food industry. *International Journal of Production Economics*, 170, 146-159.
- Gray, D. E. (2004). Doing research in the real world. Sage.
- Griffith, A., & Bhutto, K. (2009). Better environmental performance. *Management of Environmental Quality: An International Journal, 20*(5), 566-580.
- Guo, C., Shim, J. P., & Otondo, R. (2014). Social Network Services in China: An Integrated Model of Centrality, Trust, and Technology Acceptance. *Journal* of Global Information Technology Management, 13(2), 76-99.

- Hadi, N. u., Abdullah, N., & Sajilan, S. (2015). Conceptual Framework of Factors Affecting SMEs Manufacturing Business Performance. *Review of Integrated Business and Economics Research*, 4(3), 250-258.
- Haenlein, M., & Kaplan, A. M. (2004). A Beginner's Guide to Partial Least Squares Analysis. Understanding Statistics, 3(4), 283-297.
- Hair, J. F., Anderson, R. E., Tatham, R. L., & William, C. (1998). Multivariate data analysis, Upper Saddle River, NJ: Prentice Hall.
- Hair, J. F., Babin, B. J., & Krey, N. (2017). Covariance-Based Structural Equation Modeling in the Journal of Advertising: Review and Recommendations. *Journal of Advertising*, 46(1), 163-177.
- Hair, J. F., Black, W. C., Babin, B. J., Anderson, R. E., & Tatham, R. L. (1998). Multivariate Data Analysis (5th ed) Upper Saddle River.
- Hair, J. F., Ringle, C. M., & Sarstedt, M. (2011). PLS-SEM: Indeed a Silver Bullet. *The Journal of Marketing Theory and Practice*, 19(2), 139-152.
- Hair, J. F., Ringle, C. M., & Sarstedt, M. (2013). Partial Least Squares Structural Equation Modeling: Rigorous Applications, Better Results and Higher Acceptance. *Long Range Planning*, 46(1-2), 1-12.
- Hair, J. F., Risher, J. J., Sarstedt, M., & Ringle, C. M. (2018). When to use and how to report the results of PLS-SEM.
- Hair, J. F. J., Black, W. C., Babin, B. J., & Anderson, R. E. (2010). Multivariate Data Analysis, 7th edn. Prentice Hall, New Jersey.
- Hair, J. F. J., Hult, G. T. M., Ringle, C. M., & Sarstedt, M. (2014). A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM): Sage Publications.
- Hair, J. F. J., Hult, G. T. M., Ringle, C. M., & Sarstedt, M. (2017). A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM): Sage Publications.
- Hair, J. F. J., Sarstedt, M., Hopkins, L., & Kuppelwieser, V. G. (2014). Partial least squares structural equation modeling (PLS-SEM): An emerging tool in business research. *European Business Review*, 26(2), 106-121.
- Hamdoun, M., Chiappetta Jabbour, C. J., & Ben Othman, H. (2018). Knowledge transfer and organizational innovation: Impacts of quality and environmental management. *Journal of Cleaner Production*, 193, 759-770.

- Hammons, C., & Maddux, G. (1990). Total quality management in the public sector. *Management Decision*, 28(4), 15-19.
- Harnesk, R., & Abrahamsson, L. (2007). TQM: an act of balance between contradictions. *The TQM Magazine*, 19(6), 531-540.
- Henseler, J., & Chin, W. W. (2010). A Comparison of Approaches for the Analysis of Interaction Effects Between Latent Variables Using Partial Least Squares Path Modeling. *Structural Equation Modeling: A Multidisciplinary Journal*, 17(1), 82-109.
- Henseler, J., Hubona, G., & Ray, P. A. (2016). Using PLS path modeling in new technology research: updated guidelines. *Industrial Management & Data Systems*, 116(1), 2-20.
- Henseler, J., Ringle, C. M., & Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modeling. *Journal* of the Academy of Marketing Science, 43(1), 115-135.
- Henseler, J., Ringle, C. M., & Sinkovics, R. R. (2009). The Use of Partial Least Squares Path Modeling in International Marketing. *Advances in International Marketing*, 20, 277-319.
- Hietschold, N., Reinhardt, R., & Gurtner, S. (2014). Measuring critical success factors of TQM implementation successfully – a systematic literature review. *International Journal of Production Research*, 52(21), 6254-6272.
- Hodge, V., & Austin, J. (2004). A Survey of Outlier Detection Methodologies. Artificial Intelligence Review, 22(2), 85-126.
- Hoelter, J. W. (1983). The Analysis of Covariance Structures:Goodness-of-Fit Indices. SOCIOLOGICAL METHODS & RESEARCH, 11(3), 325-344.
- Hoffmann, J., Hirsch, S., & Simons, J. (2017). Identification of spatial agglomerations in the German food processing industry. *Papers in Regional Science*, 96(1), 139-162.
- Hofstede, G., Neuijen, B., Ohayv, D. D., & Sanders, G. (1990). Measuring Organizational Cultures: A Qualitative and Quantitative Study Across Twenty Cases. *Administrative service Quality*, 35(2), 286-316.
- Hu, L. T., & Bentler, P. M. (1998). Fit Indices in Covariance Structure Modeling: Sensitivity to Underparameterized Model Misspecification. *Psychol Methods*, 3(4), 424-453.

- Hulland, J., Baumgartner, H., & Smith, K. M. (2017). Marketing survey research best practices: evidence and recommendations from a review of JAMS articles. *Journal of the Academy of Marketing Science*, 46(1), 92-108.
- Hussain, S. T., Khan, U., Faheem, A., & Malik, K. Z. (2012). The Constraints to Industry in Punjab, Pakistan. *The Lahore Journal of Economics*, 17, 135-189.
- Hyder, S., & Lussier, R. N. (2016). Why businesses succeed or fail: a study on small businesses in Pakistan. *Journal of Entrepreneurship in Emerging Economies*, 8(1), 82-100.
- Ilkay, M. S., & Aslan, E. (2012). The effect of the ISO 9001 quality management system on the performance of SMEs. *International Journal of Quality & Reliability Management, 29*(7), 753-778.
- Imran, M., Hamid, S. N. b. A., & Aziz, A. b. (2018). The influence of TQM on export performance of SMEs: Empirical evidence from manufacturing sector in Pakistan using PLS-SEM. *Management Science Letters*, 483-496.
- Iqbal, N. (2013). "A framework for assessing the impact of investment in human capital development on organisational performance" PhD Thesis. University of Bedfordshire.
- Irfan, S. M., Kee, D. M. H., Qureshi, R. W., & Hussain, R. (2014). Measuring Performance of SMEs in Pakistan using PLS-SEM: Evaluating MBNQA Criteria as TQM Framework. *Science International Lahore, 26*(4), 1707-1718.
- Ismyrlis, V., Moschidis, O., & Tsiotras, G. (2015). Critical success factors examined in ISO 9001:2008-certified Greek companies using multidimensional statistics. *International Journal of Quality & Reliability Management, 32*(2), 114-131.
- Jabbour, A. B. L. d. S., Jabbour, C. J. C., Latan, H., Teixeira, A. A., & de Oliveira, J. H. C. (2015). Reprint of "Quality management, environmental management maturity, green supply chain practices and green performance of Brazilian companies with ISO 14001 certification: Direct and indirect effects". *Transportation Research Part E: Logistics and Transportation Review, 74*, 139-151.
- Jabbour, C. J. C. (2015). Environmental training and environmental management maturity of Brazilian companies with ISO14001: empirical evidence. *Journal* of Cleaner Production, 96, 331-338.

- Jabbour, C. J. C., Jabbour, A. B. L. d. S., Govindan, K., Teixeira, A. A., & Freitas,
 W. R. d. S. (2013). Environmental management and operational performance in automotive companies in Brazil: the role of human resource management and lean manufacturing. *Journal of Cleaner Production*, 47, 129-140.
- Jabeen, R. (2014). "Moderating effect of External Environment on Performance of SMEs in Pakistan" PhD Thesis. University Utara Malaysia.
- Jabeen, R., Shehu, A. M., Mahmood, R., & Bambale, A. J. a. (2015). Total Quality Management Dimensions and SME Performance: A Quantitative Approach. *IPBJ*, 7(1), 21-35.
- Jabnoun, N., & Khafaji, A. A. (2005). National Cultures for Quality Assurance and Total Quality Management. *Journal of Transnational Management*, 10(3), 3-17.
- Jackson, S. A., Gopalakrishna-Remani, V., Mishra, R., & Napier, R. (2016). Examining the impact of design for environment and the mediating effect of quality management innovation on firm performance. *International Journal* of Production Economics, 173, 142-152.
- Jarvis, C. B., Mackenzie, S. B., & Podsakoff, P. M. (2003). A Critical Review of Construct Indicators and Measurement Model Misspecification in Marketing and Consumer Research. *Journal of Consumer Research*, 30(2), 199-218.
- Jasra, J. M., Khan, D. M. A., Hunjra, A. I., Rehman, R. A. U., & Azam, D. R. I. (2011). Determinants of Business Success of Small and Medium Enterprises. *International Journal of Business and Social Science*, 2(20).
- Jayaram, J., Ahire, S. L., & Dreyfus, P. (2010). Contingency relationships of firm size, TQM duration, unionization, and industry context on TQM implementation—A focus on total effects. *Journal of Operations Management, 28*(4), 345-356.
- Jones, M. R. (2014). Identifying Critical Factors That Predict Quality Management Program Success: Data Mining Analysis of Baldrige Award Data. *Quality Management Journal*, 21(3), 49-61.
- Jong, C.-Y., Sim, A. K. S., & Lew, T. Y. (2019). The relationship between TQM and project performance: Empirical evidence from Malaysian construction industry. *Cogent Business & Management*, 6(1), 1-31.

- Jonkman, J., Bloemhof, J. M., van der Vorst, J. G. A. J., & van der Padt, A. (2017). Selecting food process designs from a supply chain perspective. *Journal of Food Engineering*, 195, 52-60.
- JUSE. (2017). Union of Japanese Scientists and Engineers: Deming Prize. Retrieved 03-01-2019, from https://www.juse.or.jp/deming_en/
- Kafetzopoulos, D., & Gotzamani, K. (2019). Investigating the role of EFQM enablers in innovation performance. *The TQM Journal*, *31*(2), 239-256.
- Kafetzopoulos, D., Gotzamani, K., & Psomas, E. (2013). Quality systems and competitive performance of food companies. *Benchmarking: An International Journal*, 20(4), 463-483.
- Kafetzopoulos, D. P., & Gotzamani, K. D. (2014). Critical factors, food quality management and organizational performance. *Food Control, 40*, 1-11.
- Kamaruddin, N. I., Khalid, F. A., & Jabar, J. (2018). Impact of Constraints towards Performance of Food Manufacturing SMEs in Malaysia. *Journal of Fundamental and Applied Sciences*, 10(6), 822-838.
- Kannan, D. (2018). Role of multiple stakeholders and the critical success factor theory for the sustainable supplier selection process. *International Journal of Production Economics*, 195, 391-418.
- Karanja, E., Zaveri, J., & Ahmed, A. (2013). How do MIS researchers handle missing data in survey-based research: A content analysis approach. *International Journal of Information Management*, 33(5), 734-751.
- Karapetrovic, S., & Willborn, W. (1998). Integration of quality and environmental management systems. *The TQM Magazine*, *10*(3), 204-213.
- Kaynak, H. (2003). The relationship between total quality management practices and their effects on firm performance. *Journal of Operations Management*, 21(4), 405-435.
- Kelloway, E. K. (1998). Using LISREL for structural equation modeling: A researcher's guide. Sage.
- Khalique, M., Bontis, N., Shaari, J. A. N. b., & Isa, A. H. M. (2015). Intellectual capital in small and medium enterprises in Pakistan. *Journal of Intellectual Capital*, 16(1), 224-238.
- Khan, B. A., & Naeem, H. (2018). Measuring the impact of soft and hard quality practices on service innovation and organisational performance. *Total Quality Management & Business Excellence, 29*(11-12), 1402-1426.

- Khan, M. A. (2011). An Empirical Study of Barriers in Implementing Total Quality Management in Service Organizations in Pakistan. Asian Journal of Business Management Studies, 2(4), 155-161.
- Khan, M. W. J., & Khalique, M. (2014). An Overview of Small and Medium Enterprises in Malaysia and Pakistan: Past, Present and Future Scenario. *Business and Management Horizons*, 2(2), 38-49.
- Khan, N. R., Awang, M., & Zulkifli, C. M. (2013). Small and Medium Enterprises and Human Resource Practices in Pakistan. *International Journal of Asian Social Science*, 3(2), 460-471.
- Kharub, M., & Sharma, R. K. (2016). Investigating the role of CSF's for successful implementation of quality management practices in MSMEs. *International Journal of System Assurance Engineering and Management*, 7(1), 247-273.
- Khattak, J. K., Arslan, M., & Umair, M. (2011). SMEs' export problems in Pakistan. Journal of Business Management and Economics, 2(5), 192-199.
- Khurshid, M. A., Amin, M., & Ismail, W. K. W. (2018). Total quality and socially responsible management (TQSR-M):an integrated conceptual framework. *Benchmarking: An International Journal*, 25(8), 2566-2588.
- Kianpour, K., Jusoh, A., Mardani, A., Streimikiene, D., Cavallaro, F., Nor, K., & Zavadskas, E. (2017). Factors Influencing Consumers' Intention to Return the End of Life Electronic Products through Reverse Supply Chain Management for Reuse, Repair and Recycling. *Sustainability*, 9(9), 1657.
- King, A., & Lenox, M. (2002). Exploring the Locus of Profitable Pollution Reduction. *Management Science*, 48(2), 289-299.
- Klare, G. R. (1950). Understandability and indefinite answers to public opinion questions. *International Journal of Opinion and Attitude Research*, 4(1), 91-96.
- Kock, N. (2015). One-tailed or two-tailed P values in PLS-SEM? International Journal of e-Collaboration, 11(2), 1-7.
- Kumar, R. (2011). *Research methodology a step-by-step guide for begineers*: SAGE Publication.
- Kureshi, N., Qureshi, F., & Sajid, A. (2010). Current health of quality management practices in service sector SME: A case study of Pakistan. *The TQM Journal*, 22(3), 317-329.

- Kureshi, N. I., Mann, R., Khan, M. R., & Qureshi, M. F. (2009). Quality Management Practices of SME in Developing Countries: A Survey of Manufacturing SME in Pakistan. *Journal of Quality and Technology Management*, 5(2), 63-89.
- Laari, S., Töyli, J., Solakivi, T., & Ojala, L. (2016). Firm performance and customerdriven green supply chain management. *Journal of Cleaner Production*, 112, 1960-1970.
- Ladhari, R. (2012). The lodging quality index: an independent assessment of validity and dimensions. *International Journal of Contemporary Hospitality Management*, 24(4), 628-652.
- Lakhal, L., Pasin, F., & Limam, M. (2006). Quality management practices and their impact on performance. *International Journal of Quality & Reliability Management*, 23(6), 625-646.
- Lannelongue, G., Gonzalez-Benito, J., Gonzalez-Benito, O., & Gonzalez-Zapatero,
 C. (2015). Time compression diseconomies in environmental management:
 The effect of assimilation on environmental performance. *J Environ Manage*, 147, 203-212.
- Latif, Y., Fiaz, M., & Shoaib, M. (2014). Important TQM Implementation Contributors in Pakistani Petrochemical Sector. *Pak.j.stat.oper.res.*, 10(3), 331-348.
- Latif, Y., Iqbal, P., & Rehman, S. u. (2014). Important TQM Implementation Contributors in Service oriented Organization like Banks in Pakistan. *Management Research Report*, 2(5), 133-199.
- Lau, R. S. M., Zhao, X., & Xiao, M. (2004). Assessing quality management in China with MBNQA criteria. *International Journal of Quality & Reliability Management*, 21(7), 699-713.
- Lee, S. M. (2015). The age of quality innovation. *International Journal of Quality Innovation, 1*(1), 1-5.
- Lefebvre, V. M., De Steur, H., & Gellynck, X. (2015). External sources for innovation in food SMEs. *British Food Journal*, 117(1), 412-430.
- Leonidou, L. C., Leonidou, C. N., Fotiadis, T. A., & Zeriti, A. (2013). Resources and capabilities as drivers of hotel environmental marketing strategy: Implications for competitive advantage and performance. *Tourism Management*, 35, 94-110.

- Leung, S.-O. (2011). A Comparison of Psychometric Properties and Normality in 4-,
 5-, 6-, and 11-Point Likert Scales. *Journal of Social Service Research*, 37(4),
 412-421.
- Li, D., Zheng, M., Cao, C., Chen, X., Ren, S., & Huang, M. (2017). The impact of legitimacy pressure and corporate profitability on green innovation: Evidence from China top 100. *Journal of Cleaner Production*, 141, 41-49.
- Link, S., & Naveh, E. (2006). Standardization and Discretion: Does the Environmental Standard ISO 14001 Lead to Performance Benefits? *IEEE Transactions on Engineering Management*, 53(4), 508-519.
- Liu, Y.-L., & Ko, P.-F. (2018). A modified EFQM Excellence Model for effective evaluation in the hotel industry. *Total Quality Management & Business Excellence*, 29(13-14), 1580-1593.
- Llach, J., Perramon, J., Alonso-Almeida, M. d. M., & Bagur-Femenías, L. (2013). Joint impact of quality and environmental practices on firm performance in small service businesses: an empirical study of restaurants. *Journal of Cleaner Production, 44*, 96-104.
- Lohmöller, J.-B. (2013). Latent Variable Path Modelling with Partial Least Squares: Springer Science & Business Media.
- López-Gamero, M. D., Molina-Azorín, J. F., & Claver-Cortes, E. (2011). The relationship between managers' environmental perceptions, environmental management and firm performance in Spanish hotels: a whole framework. *International Journal of Tourism Research*, 13(2), 141-163.
- Lynn, M. R. (1986). Determination and quantification of content validity. Nursing Research, 35(6), 382-385.
- Ma, M., & Agarwal, R. (2007). Through a Glass Darkly: Information Technology Design, Identity Verification, and Knowledge Contribution in Online Communities. *Information Systems Research*, 18(1), 42-67.
- Mahmud, N., & Hilmi, M. F. (2014). TQM and Malaysian SMEs Performance: The Mediating Roles of Organization Learning. *Procedia - Social and Behavioral Sciences*, 130, 216-225.
- Majid, M. S. A., Basri, H., Nopita, E., & Fahlevi, H. (2016). The Effect of Organizational Culture, Leadership Style, and Functional Position on Organizational Commitment and Their Impact on the Performance of Internal

Auditors in Aceh, Indonesia. *BRAND*. *Broad Research in Accounting*, *Negotiation, and Distribution*, 7(1), 37-50.

- Manan, F. B. A. (2006). "Achieving Sustainable Food Industries: Good Practices in Environmental Management", Master's Thesis,. Universiti Teknologi Malaysia.
- Mardani, A., Jusoh, A., Nor, K., Kish Hazrat Soltan, E., & Kazemilari, M. (2013). Total Quality Management and Organizational Culture framework for Small and Medium-Sized Businesses (ISMBs) in Iran. *Caspian Journal of Applied Sciences Research*, 2(10), 43-60.
- Mardani, A., Jusoh, A., Soltan, E. K. H., & Bagheri, M. M. (2012). The Effect of Organizational Culture Stemming from National Culture on Total Quality Management Values, Case Study: Iranian Small and Medium Enterprises (SME's). Paper presented at the International Conference on Management, Economics and Finance (ICMEF 2012), Malaysia.
- Massoud, M. A., Fayad, R., El-Fadel, M., & Kamleh, R. (2010). Drivers, barriers and incentives to implementing environmental management systems in the food industry: A case of Lebanon. *Journal of Cleaner Production*, 18(3), 200-209.
- Mellat-Parast, M. (2015). A longitudinal assessment of the linkages among the Baldrige criteria using independent reviewers' scores. *International Journal* of Production Economics, 164, 24-34.
- Mellat-Parast, M., Adams, S. G., & Jones, E. C. (2011). Improving operational and business performance in the petroleum industry through quality management. *International Journal of Quality & Reliability Management, 28*(4), 426-450.
- Melnyk, S. A., Sroufe, R. P., & Calantone, R. (2003). Assessing the impact of environmental management systems on corporate and environmental performance. *Journal of Operations Management*, 21(3), 329-351.
- Memon, M. A., Ting, H., Ramayah , T., Chuah, F., & Cheah, J.-H. (2017). Editorial -A Review of the Methodological Misconceptions and Guidelines related to the Application of Structural Equation Modeling: A Malaysian Scenario. *Journal of Applied Structural Equation Modeling*, 1(1), 1-8.
- Mezinska, I., & Strode, S. (2015). Emerging Horizons of Environmental Management in Food Sector Companies. *Procedia - Social and Behavioral Sciences*, 213, 527-532.

- Miles, M. P., & Covin, J. G. (2000). Environmental Marketing: A Source of Reputational, Competitive, and Financial Advantage. *Journal of Business Ethics*, 23(3), 299-311.
- Mitchell, M. L., & Jolley, J. M. (2012). Research design explained: Cengage Learning.
- Modgil, S., & Sharma, S. (2016). Total productive maintenance, total quality management and operational performance: An empirical study of Indian pharmaceutical industry. *Journal of Quality in Maintenance Engineering*, 22(4), 353-377.
- MOF. (2017a). Agriculture Sector. Retrieved 15-05-2017, from http://www.finance.gov.pk/survey/chapters 16/02 Agriculture.pdf
- MOF. (2017b). Manufacturing Sector. Retrieved 15-05-2017, from http://www.finance.gov.pk/survey/chapters 16/03 Manufacturing.pdf
- MOF. (2018). Ministry of Finance: Popolation, labour force, and employment. Retrieved 15-12-2018, from <u>http://www.finance.gov.pk/survey/chapters_18/12-Population.pdf</u>
- Mohammad, J., Quoquab, F., Makhbul, Z. M., & Ramayah, T. (2016). Bridging the gap between justice and citizenship behavior in Asian culture. *Cross Cultural* & Strategic Management, 23(4), 633-656.
- Mohammad, M., Osman, M. R., Yusuff, R. M., & Ismail, N. (2006). Strategies and critical success factors for integrated management systems implementation.
 Paper presented at the 35th International Conference on Computers and Industrial Engineering.
- Mohd Fuzi, N., Habidin, N. F., Janudin, S. E., & Ong, S. Y. Y. (2018). Critical success factors of environmental management accounting practices: findings from Malaysian manufacturing industry. *Measuring Business Excellence*, 0(0), null.
- Molina-Azorín, J. F., Claver-Cortés, E., Pereira-Moliner, J., & Tarí, J. J. (2009). Environmental practices and firm performance: an empirical analysis in the Spanish hotel industry. *Journal of Cleaner Production*, 17(5), 516-524.
- Molina-Azorín, J. F., Tarí, J. J., Claver-Cortés, E., & López-Gamero, M. D. (2009). Quality management, environmental management and firm performance: A review of empirical studies and issues of integration. *International Journal of Management Reviews*, 11(2), 197-222.

- Molina-Azorín, J. F., Tarí, J. J., Pereira-Moliner, J., López-Gamero, M. D., & Pertusa-Ortega, E. M. (2015). The effects of quality and environmental management on competitive advantage: A mixed methods study in the hotel industry. *Tourism Management*, 50, 41-54.
- Moray, N. (2015). Culture, Context and Performance. *Cultural Ergonomics.*, *4*, 31-59.
- Moumen, M., & El Aoufir, H. (2017). Quality, safety and environment management systems (QSE): analysis of empirical studies on integrated management systems (IMS). *Journal of Decision Systems*, 1-22.
- MPDR. (2016). Pakistan Vision 2025. Retrieved 10-03-2017, from http://pc.gov.pk/vision/visiondoc
- Muhammad, L., Mahadi, B., & Hussin, N. (2017). Influence of social capital on customer's relationship satisfaction in the Pakistani banking industry. Asia Pacific Journal of Marketing and Logistics, 29(5), 1036-1054.
- Nadeem, O., & Hameed, R. (2008). Evaluation of environmental impact assessment system in Pakistan. *Environmental Impact Assessment Review*, 28(8), 562-571.
- Nair, A. (2006). Meta-analysis of the relationship between quality management practices and firm performance—implications for quality management theory development. *Journal of Operations Management, 24*(6), 948-975.
- Najib, M., & Kiminami, A. (2011). Innovation, cooperation and business performance. *Journal of Agribusiness in Developing and Emerging Economies*, 1(1), 75-96.
- Naor, M., Goldstein, S. M., Linderman, K. W., & Schroeder, R. G. (2008). The Role of Culture as Driver of Quality Management and Performance: Infrastructure Versus Core Quality Practices*. *Decision Sciences*, 39(4), 671-702.
- Nawaz, M., Mohd Harif, M. A. A., & Azhari, A. B. (2018). The non-financial determinants of the financial derivatives' usage within SME businesses in Pakistan. Advances in Social Sciences Research Journal, 5(9).
- Nazir, N. A., & Lone, M. A. (2008). Validation of Denison's Model of Organisational Culture and Effectiveness in the Indian Context. *Vision*, 12(1), 49-58.

- Neely, A., Gregory, M., & Platts, K. (1995). Performance measurement system design: A literature review and research agenda. *International Journal of Operations & Production Management*, 15(4), 80-116.
- Nguyen, N. T. D., & Aoyama, A. (2015). The impact of cultural differences on technology transfer: Management practice moderation. *Journal of Manufacturing Technology Management*, 26(7), 926-954.
- NIST. (2017). National Institute of Standards and Technology: Malcolm Baldridge National Quality Awards (MBNQA). Retrieved 11-01-2019, from https://www.nist.gov/topics/baldrige-framework-criteria
- NIST. (2018). National Institute of Standards and Technology. Retrieved 10-10-2018, from https://www.nist.gov/sites/default/files/documents/2017/02/09/2017-2018baldrige-excellence-builder.pdf
- Nor, N. G. M., Bhuiyan, A. B., Said, J., & Alam, S. S. (2016). Innovation barriers and risks for food processing SMEs in Malaysia: A logistic regression analysis. *Malaysian Journal of Society and Space*, 12(2), 167-178.
- Nordin, N., Ashari, H., & Rajemi, M. F. (2014). A Case Study of Sustainable Manufacturing Practices. *Journal of Advanced Management Science*, 2(1), 12-16.
- Notarnicola, B., Tassielli, G., Renzulli, P. A., Castellani, V., & Serenella, S. (2016). Environmental impacts of food consumption in Europe. *Journal of Cleaner Production*, 1-13.
- NPO. (2017). National Productivity Organization: Prime Minister Quality Award. Retrieved 15-10-2018, from <u>http://www.npo.gov.pk/prime-minister-quality-award/</u>
- Nunhes, T. V., Motta, L. C. F., & Oliveira, O. J. d. (2016). Evolution of integrated management systems research on the Journal of Cleaner Production: Identification of contributions and gaps in the literature. *Journal of Cleaner Production, 139*, 1234-1244.
- Nunnaly, J. C. (1978). Psychometric Theory. 2nd ed.: McGraw-Hill, New York.
- O'Reilly, S., Kumar, A., & Adam, F. (2015). The role of hierarchical production planning in food manufacturing SMEs. *International Journal of Operations & Production Management, 35*(10), 1362-1385.

- Ofori, G., Gang, G., & Briffett, C. (2002). Implementing environmental management systems in construction: lessons from quality systems. *Building and Environment*, *37*(12), 1397-1407.
- Ogbeibu, S., Senadjki, A., & Gaskin, J. (2018). The moderating effect of benevolence on the impact of organisational culture on employee creativity. *Journal of Business Research, 90*, 334-346.
- Oldfield, T. L., White, E., & Holden, N. M. (2016). An environmental analysis of options for utilising wasted food and food residue. *Journal of Environment Management*, 183(Pt 3), 826-835.
- Oliveira, O. J. d., Serra, J. R., & Salgado, M. H. (2010). Does ISO 14001 work in Brazil? *Journal of Cleaner Production*, 18(18), 1797-1806.
- Ortolano, L., Sanchez-Triana, E., Afzal, J., Ali, C. L., & Rebellón, S. A. (2014). Cleaner production in Pakistan's leather and textile sectors. *Journal of Cleaner Production*, 68, 121-129.
- Oskarsson, K., & Malmborg, F. v. (2005). Integrated Management Systems as a Corporate Response to Sustainable Development. *Corp. Soc. Responsib. Environ. Mgmt.*, 12, 121-128.
- Pallant, J. (2011). A step by step guide to data analysis using the SPSS program, Survival Manual: McGraw-Hill Education (UK).
- Patyal, V. S., & Koilakuntla, M. (2017). The impact of quality management practices on performance: an empirical study. *Benchmarking: An International Journal*, 24(2), 511-535.
- Patyal, V. S., & Koilakuntla, M. (2018). Impact of organizational culture on quality management practices: an empirical investigation. *Benchmarking: An International Journal*, 25(5), 1406-1428.
- Peng, D. X., & Lai, F. (2012). Using partial least squares in operations management research: A practical guideline and summary of past research. *Journal of Operations Management*, 30(6), 467-480.
- Pereira-Moliner, J., Claver-Cortés, E., Molina-Azorín, J. F., & José Tarí, J. (2012). Quality management, environmental management and firm performance: direct and mediating effects in the hotel industry. *Journal of Cleaner Production, 37*, 82-92.
- Perneger, T. V., Courvoisier, D. S., Hudelson, P. M., & Gayet-Ageron, A. (2015). Sample size for pre-tests of questionnaires. *Qual Life Res, 24*(1), 147-151.

- Pertusa-Ortega, E. M., López-Gamero, M. D., Pereira-Moliner, J., Tarí, J.-J., & Molina-Azorín, J. F. (2018). Antecedents of Environmental Management: The Influence of Organizational Design and Its Mediating Role Between Quality Management and Environmental Management. *Organization & Environment*, 31(4), 425-443.
- Pinho, J. C., Rodrigues, A. P., & Dibb, S. (2014). The role of corporate culture, market orientation and organisational commitment in organisational performance: The case of non-profit organisations. *Journal of Management Development*, 33(4), 374-398.
- Pipatprapa, A., Huang, H.-H., & Huang, C.-H. (2016). A Novel Environmental Performance Evaluation of Thailand's Food Industry Using Structural Equation Modeling and Fuzzy Analytic Hierarchy Techniques. *Sustainability*, 8(246), 2-16.
- Pira, B., & Zylfijaj, R. (2017). Assessment of critical success factors of TQM culture in hospitality sector in Kosovo. *International Journal of Research in Business* and Social Science, 6(1), 65-77.
- Podsakoff, P. M., MacKenzie, S. B., & Lee, J.-Y. (2003). Common method biases in behavioral research: a critical review of the literature and recommended remedies. *Journal of Applied Psychology*, 88(5), 879-903.
- Podsakoff, P. M., MacKenzie, S. B., & Podsakoff, N. P. (2012). Sources of method bias in social science research and recommendations on how to control it. *Annu Rev Psychol*, 63, 539-569.
- Prajogo, D. I., & McDermott, C. M. (2005). The relationship between total quality management practices and organizational culture. *International Journal of Operations & Production Management*, 25(11), 1101-1122.
- Prajogo, D. I., & McDermott, C. M. (2011). The relationship between multidimensional organizational culture and performance. *International Journal of Operations & Production Management*, 31(7), 712-735.
- Psomas, E., Vouzas, F., & Kafetzopoulos, D. (2014). Quality management benefits through the "soft" and "hard" aspect of TQM in food companies. *The TQM Journal*, *26*(5), 431-444.
- Psomas, E. L., & Fotopoulos, C. V. (2010). Total quality management practices and results in food companies. *International Journal of Productivity and Performance Management*, 59(7), 668-687.

- Pun, K. F., Fung, Y. K., & Wong, F. Y. (1998). Identification of critical factors for total quality environmental management. Proceedings of the 3rd Annual International Conference on Industrial Engineering Theories, Application and Practice. Hong Kong, December, PN114, pp. 1-9.
- Pun, K. F., Hui, I. K., Lau, H. C. W., Law, H. W., & Lewis, W. G. (2002). Development of an EMS planning framework for environmental management practices. *International Journal of Quality & Reliability Management*, 19(6), 688-709.
- Punch, K. F. (2013). Introduction to social research: Quantitaive and qualitative approaches: Sage.
- Punjab Skill Development Fund. (2015). Food Processing Sector Skills Study. Retrieved 16-12-2017, from https://psdf.org.pk/wpcontent/uploads/2015/12/Food-Processing-Report-1-1.pdf
- Quader, M. S., Kamal, M. M., & Hassan, A. B. M. E. (2016). Sustainability of positive relationship between environmental performance and profitability of SMEs. *Journal of Enterprising Communities*, 10(2), 138-163.
- Quazi, H. A. (1999). Implementation of an environmental management system: the experience of companies operating in Singapore. *Industrial Management and Data Systems*, 99(7), 302-311.
- Quazi, H. A., & Padibjo, S. R. (1998). A journey toward total quality management through ISO 9000 certification - a study on small- and medium-sized enterprises in Singapore. *International Journal of Quality & Reliability Management*, 15(5), 489-508.
- Rahman, S. u. (2001). A comparative study of TQM practice and organisational performance of SMEs with and without ISO 9000 certification. *International Journal of Quality & Reliability Management, 18*(1), 35-49.
- Ram, J., & Corkindale, D. (2014). How "critical" are the critical success factors (CSFs)? Examining the role of CSFs for ERP *Business Process Management Journal*, 20(1), 151-174.
- Rao, P., & Holt, D. (2005). Do green supply chains lead to competitiveness and economic performance? *International Journal of Operations & Production Management*, 25(9), 898-916.

- Rashid, F., & Fazal, N. (2017). Study of environmental indicators in a thermal power plant in Lahore, Pakistan. *Management of Environmental Quality: An International Journal*, 28(6), 930-945.
- Rasi, R. Z. R. M., Abdekhodaee, A., & Nagarajah, R. (2014). Stakeholders' involvements in the implementation of proactive environmental practices. *Management of Environmental Quality: An International Journal, 25*(2), 132-149.
- Rebelo, M. F., Santos, G., & Silva, R. (2016). Integration of management systems: towards a sustained success and development of organizations. *Journal of Cleaner Production*, 127, 96-111.
- Rehman, N. U. (2016). Network alliances and firms' performance: a panel data analysis of Pakistani SMEs. *Eurasian Business Review*, 6(1), 37-52.
- Reinartz, W., Haenlein, M., & Henseler, J. (2009). An empirical comparison of the efficacy of covariance-based and variance-based SEM. *International Journal* of Research in Marketing, 26(4), 332-344.
- Rezaei, G., Mardani, A., Senin, A. A., Wong, K. Y., Sadeghi, L., Najmi, M., & Shaharoun, A. M. (2018). Relationship between culture of excellence and organisational performance in Iranian manufacturing companies. *Total Quality Management & Business Excellence, 29*(1-2), 94-115.
- Rezaei, S. (2015). Segmenting consumer decision-making styles (CDMS) toward marketing practice: A partial least squares (PLS) path modeling approach. *Journal of Retailing and Consumer Services*, 22, 1-15.
- Rogo, H. B., Shariff, M. N. M., & Hafeez, M. H. (2018). Investigating the Effect of Entrepreneurship Development on the Relationship between Market Orientation, Total Quality Management and SMEs Performance in Kano. *International Journal of Supply Chain Management*, 7 (2), 107-115.
- Rollenhagen, C., & Wahlstrom, B. (2007, 26-31 Aug. 2007). Management systems and safety culture; reflections and suggestions for research. Paper presented at the 2007 IEEE 8th Human Factors and Power Plants and HPRCT 13th Annual Meeting.
- Roni, M. S. M. (2015). "An analysis of insider dysfunctional behavours in an accounting information system environment" PhD Thesis. Edith Cowan University.

- Rowley, J. (2014). Designing and using research questionnaires. *Management Research Review*, 37(3), 308-330.
- Ruane, J. M. (2005). Essentials of Research Methods: A Guide to Social Science Research: Oxford, Blackwell Publishers.
- Rusko, M., Sablik, J., Marková, P., Lach, M., & Friedrich, S. (2014). Sustainable Development, Quality Management System and Environmental Management System in Slovak Republic. *Procedia Engineering*, 69, 486-491.
- Russo, M. V., & Fouts, P. A. (1997). A Resource-Based Perspective On Corporate Environmental Performance And Profitability. *Academy of Management Journal*, 40(3), 534-559.
- Saad, A. M. A. (2016). "An Investigation into the Implementation of Total Quality Environmental Management (TQEM) for Sustainability in Libyan Food Industry" PhD Thesis. Nottingham Trent University, UK.
- Saad, R., & Asaad, M. N. M. (2015). Does organizational culture moderate the relationship between ISO 9000 soft factors and organizational performance? *The Journal of Developing Areas, 49*(3), 379-394.
- Sadikoglu, E., & Olcay, H. (2014). The effects of total quality management practices on performance and the reasons of and the barriers to TQM practices in Turkey. *Advances in Decision Sciences*, 1-17.
- Saha, S., & Kumar, S. P. (2018). Organizational culture as a moderator between affective commitment and job satisfaction: Empirical evidence from Indian public sector enterprises. *International Journal of Public Sector Management*, 31(2), 184-206.
- Sahoo, S., & Yadav, S. (2018). Total Quality Management in Indian Manufacturing SMEs. *Procedia Manufacturing*, *21*, 541-548.
- Sala, S., Anton, A., McLaren, S. J., Notarnicola, B., Saouter, E., & Sonesson, U. (2016). In quest of reducing the environmental impacts of food production and consumption. *Journal of Cleaner Production*.
- Salaheldin, S. I. (2009). Critical success factors for TQM implementation and their impact on performance of SMEs. *International Journal of Productivity and Performance Management*, 58(3), 2015-2237.
- Saleh, R. A., Sweis, R. J., & Mahmoud Saleh, F. I. (2018). Investigating the impact of hard total quality management practices on operational performance in

manufacturing organizations: Evidence from Jordan. *Benchmarking: An International Journal, 25*(7), 2040-2064.

- Sambasivan, M., & Fei, N. Y. (2008). Evaluation of critical success factors of implementation of ISO 14001 using analytic hierarchy process (AHP): a case study from Malaysia. *Journal of Cleaner Production*, 16(13), 1424-1433.
- Sampaio, P., Saraiva, P., & Monteiro, A. (2012). A comparison and usage overview of business excellence models. *The TQM Journal*, 24(2), 181-200.
- Samson, D., & Terziovski, M. (1999). The relationship between total quality management practices and operational performance. *Journal of Operations Management*, *17*, 393-409.
- Sánchez-Triana, E., Biller, D., Nabi, I., Ortolano, L., Dezfuli, G., Afzal, J., & Enriquez, S. (2014). *Revitalizing Industrial Growth in Pakistan*: WORLD BANK GROUP Washington, D.C.
- Santos, G., Barros, S., Mendes, F., & Lopes, N. (2013). The main benefits associated with health and safety management systems certification in Portuguese small and medium enterprises post quality management system certification. *Safety Science*, *51*(1), 29-36.
- Santos, G., Mendes, F., & Barbosa, J. (2011). Certification and integration of management systems: the experience of Portuguese small and medium enterprises. *Journal of Cleaner Production*, 19(17-18), 1965-1974.
- Santos, G., Rebelo, M., Lopes, N., Alves, M. R., & Silva, R. (2015). Implementing and certifying ISO 14001 in Portugal: motives, difficulties and benefits after ISO 9001 certification. *Total Quality Management & Business Excellence*, 27(11-12), 1211-1223.
- Saraph, J. V., Benson, P. G., & Schroeder, R. G. (1989). An Instrument for Measuring the Critical Factors of Quality Management. *Decision Sciences*, 20(4), 810-829.
- Saunders, M., Lewis, P., & Thornhill, A. (2003). Research Methods for Business Students: Pearson Education Limited
- Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research methods for business students* (5th edition ed.): Pearsons.
- Saunders, N. K. (2011). Research methods for business students, 5/e: Pearson Education India.

- Schroeder, R. G., Linderman, K., & Zhang, D. (2005). Evolution of Quality: First Fifty Issues of Production and Operations Management. *Production and Operations Management*, 14(4), 468-481.
- Schumacker, R., & Lomax, R. (2010). Structural equation modeling, Routledge, Taylor & Francis Group, New York.
- Schumacker, R. E., & Lomax, R. G. (1996). *A beginner's guide to structural equation modeling. Mahwah, New Jersey.* : Lawrence Erlbaum Associates.
- Sekaran, U. (2003). *Rerearch Methods for Business: A Skill-Building Approach* (Fourth edition ed.): John Wiley & Sons.
- Sekaran, U., & Bougie, R. (2016). *Research methods for business: A skill building approach*: John Wiley & Sons.
- Shafiq, M. (2011). "An Investigation of Total Quality Management Practices in Pakistan" PhD Thesis, . University of York.
- Shafiq, M., Lasrado, F., & Hafeez, K. (2017). The effect of TQM on organisational performance: empirical evidence from the textile sector of a developing country using SEM. *Total Quality Management & Business Excellence*, 1-22.
- Sharma, V. K., Chandna, P., & Bhardwaj, A. (2017). Green supply chain management related performance indicators in agro industry: A review. *Journal of Cleaner Production*, 141, 1194-1208.
- Shen, X. X., Tan, K. C., & Xie, M. (2000). An integrated approach to innovative product development using Kano's model and QFD. *European Journal of Innovation Management*, 3(2), 91-99.
- Shokri, Alireza, & Nabhani, F. (2018). Quality management vision of future early career Operations Managers: a model approach. *International Journal of Quality & Reliability Management*, 1-33.
- Sila, I. (2007). Examining the effects of contextual factors on TQM and performance through the lens of organizational theories: An empirical study. *Journal of Operations Management*, 25(1), 83-109.
- Sila, I. (2018). Country and sector effects on the relationships among TQM practices and key performance measures. *International Journal of Productivity and Performance Management*, 67(8), 1371-1393.
- Sila, I., & Ebrahimpour, M. (2005). Critical linkages among TQM factors and business results. International Journal of Operations & Production Management, 25(11), 1123-1155.

- Simon, A., Bernardo, M., Karapetrovic, S., & Casadesus, M. (2013). Implementing integrated management systems in chemical firms. *Total Quality Management & Business Excellence*, 24(3-4), 294-309.
- Simon, A., Karapetrovic, S., & Casadesús, M. (2012). Difficulties and benefits of integrated management systems. *Industrial Management & Data Systems*, 112(5), 828-846.
- Singh, N., Jain, S., & Sharma, P. (2015). Motivations for implementing environmental management practices in Indian industries. *Ecological Economics*, 109, 1-8.
- Singh, P. J., Dean, C. M. W., & Chee-Chuong, S. (2013). Deining Management Method: Subjecting Theory to Moderating and Contextual Effects. *Quality Management Journal*, 20(3), 41-69.
- Singh, V., & Koilakuntla, P. M. (2017). The impact of quality management practices on performance: an empirical study *Benchmarking: An International Journal*, 24(2).
- Sinha, N., & Dhall, N. (2018). Mediating effect of TQM on relationship between organisational culture and performance: evidence from Indian SMEs. *Total Quality Management & Business Excellence*, 1-25.
- Sinha, N., Garg, A. K., & Dhall, N. (2016). Effect of TQM principles on performance of Indian SMEs: the case of automotive supply chain. *The TQM Journal*, 28(3), 338-359.
- Siva, V., Gremyr, I., Bergquist, B., Garvare, R., Zobel, T., & Isaksson, R. (2016). The support of Quality Management to sustainable development: a literature review. *Journal of Cleaner Production*, 138, 148-157.
- SMEDA.
 (2007).
 SME
 Policy.
 Retrieved
 18-3-2017,
 from

 http://www.smeda.org/index.php?option=com_content&view=article&id=58:sme-policy-development&catid=2
- SMEDA. (2009). SME Sector Genesis, Challenges and Prospects. Retrieved 25-12-2017, from https://www.smeda.org/index.php?option=com_phocadownload&view=categ ory&id=46:smeda-publications&Itemid=566
- SMEDA. (2010a). Draft Report Industrial Clusters (Punjab):Investor's Guide.Retrieved15-2-2017,from

https://smeda.org/index.php?option=com_content&view=article&id=69&Ite mid=174

- SMEDA. (2010b). World Food Market and Compliance Requirements Guidelines for Food Industry. Retrieved 15-3-2017, from <u>http://www.smeda.org/index.php?option=com_content&view=article&id=69</u> <u>&Itemid=174</u>
- SMEDA. (2017). Small and Medium Enterprise Development Authority. from https://smeda.org/
- Sohail, M. S., & Hoong, T. B. (2003). TQM practices and organizational performances of SMEs in Malaysia: Some empirical observations. *Benchmarking: An International Journal*, 10(1), 37-53.
- Sokro, E. (2012). Analysis of the relationship that exists between organizational culture, motivation and performance. *Problems of Management in the 21st Century*, *3*(106-119).
- Spector, P. E. (2006). Method Variance in Organizational Research:Truth or Urban Legend? *Organizational Research Methods*, 9(2), 221-232.
- Spencer, S. Y., Adams, C., & Yapa, P. W. S. (2013). The mediating effects of the adoption of an environmental information system on top management's commitment and environmental performance. *Sustainability Accounting, Management and Policy Journal,* 4(1), 75-102.
- Sroufe, R. (2003). Effect of Environmental Management Systems on Environmental Management Practices and Operations. *Production and Operations Management*, 12(3), 416-431.
- State Bank of Pakistan. (2017). The growth of the processed food industry in Pakistan: Changing trends and key challenges.
- Sumaedi, S., & Yarmen, M. (2015). The Effectiveness of ISO 9001 Implementation in Food Manufacturing Companies: A Proposed Measurement Instrument. *Procedia Food Science*, 3, 436-444.
- Switch Asia. (2016). National Roundtable on Sustainable Consumption and Production (SCP). Retrieved 22-01-2017, from <u>http://www.switch-asia.eu/events/national-roundtable-on-sustainable-consumption-and-production-scp-national-action-plan-for-pakista/</u>
- Tabachnick, G. B., & Fidell, L. S. (2007). Using Multivariant Statistics: Pearson Publications.

- Talib, F., & Rahman, Z. (2010). Critical Success Factors of TQM in Service Organizations: A Proposed Model. Services Marketing Quarterly, 31(3), 363-380.
- Talib, H. H. A., Ali, K. A. M., & Idris, F. (2013). Quality management framework for the SME's food processing industry in Malaysia. *International Food Research Journal*, 20(1), 147-164.
- Talib, H. H. A., Ali, K. A. M., & Idris, F. (2014). Critical success factors of quality management practices among SMEs in the food processing industry in Malaysia. *Journal of Small Business and Enterprise Development*, 21(1), 152-176.
- Tangen, S. (2005). Demystifying productivity and performance. *International Journal of Productivity and Performance Management*, 54(1), 34-46.
- Tanninen, K., Puumalainen, K., & Sandström, J. (2010). The power of TQM: analysis of its effects on profitability, productivity and customer satisfaction. *Total Quality Management & Business Excellence*, 21(2), 171-184.
- Tarí, J. J., & Molina-Azorín, J. F. (2010). Integration of quality management and environmental management systems. *The TQM Journal*, 22(6), 687-701.
- Tari, J. J., Molina-Azorin, J. F., & Heras, I. (2012). Benefits of the ISO 9001 and ISO 14001 standards: A literature review. *Journal of Industrial Engineering* and Management, 297-322.
- Tarí, J. J., Molina, J. F., & Castejón, J. L. (2007). The relationship between quality management practices and their effects on quality outcomes. *European Journal of Operational Research*, 183(2), 483-501.
- Tari', J. J., Claver-Corte' s, E., Pereira-Moliner, J., & F., M.-A. n. (2010). Levels of quality and environmental management in the hotel industry: Their joint influence on firm performance. *International Journal of Hospitality Management, 29*, 500-510.
- Tayeb, M. (1994). Organizations and national culture: Methodology considered. Organization studies, 15(3), 429-445.
- Tehseen, S., Sajilan, S., Gadar, K., & Ramayah, T. (2017). Assessing Cultural Orientation as a Reflective-Formative Second Order Construct - A Recent PLS-SEM Approach. *Review of Integrative Business and Economics Research*, 6(2), 38-63.
- Teijlingen, E. V., & Hundley, V. (2002). The importance of pilot studies. *Nursing Standard*, *16*(40), 33-36.
- Teixeira, A. A., Jabbour, C. J. C., Latan, H., de Oliveira, J. H. C., Freitas, W. R. d. S., & Teixeira, T. B. (2017). The importance of quality management for the effectiveness of environmental management: evidence from companies located in Brazil. *Total Quality Management & Business Excellence*, 1-12.
- Tenenhaus, M., Vinzi, V. E., Chatelin, Y.-M., & Lauro, C. (2005). PLS path modeling. *Computational Statistics & Data Analysis*, 48(1), 159-205.
- The Tribune. (2013). Only 2% of businesses are registered in Pakistan. Retrieved 14-08-2018, from https://tribune.com.pk/story/519331/only-2-of-businesses-are-registered-in-pakistan/
- Tian, M., Deng, P., Zhang, Y., & Salmador, M. P. (2018). How does culture influence innovation? A systematic literature review. *Management Decision*, 56(5), 1088-1107.
- Tjärnemo, H., & Södahl, L. (2015). Swedish food retailers promoting climate smarter food choices—Trapped between visions and reality? *Journal of Retailing and Consumer Services*, 24, 130-139.
- Török, Á., Tóth, J., & Balogh, J. M. (2018). Push or Pull? The nature of innovation process in the Hungarian food SMEs. *Journal of Innovation & Knowledge*.
- Trade Related Technical Assistance. (2012). Rejection of export consignments Retrieved 01-11-2018, from <u>http://trtapakistan.org/oldsite/archives/3261</u>
- Trading Economics. (2018). Pakistan Exports. Retrieved 15-12-2018, from https://tradingeconomics.com/pakistan/exports
- Tran, V. M., & Perry, J. A. (2003). Challenges to using neem (Azadirachta indica var. sianensis Valenton) in Thailand. *Economic Botany*, 57(1), 93-102.
- TRTAP. (2014). Sanitary and phyto-sanitary compliance. Retrieved 22-11-2018, from <u>http://trtapakistan.org/wp-content/uploads/2013/10/SPS-Final-February-2014.pdf</u>
- Tung, A., Baird, K., & Schoch, H. (2014a). The association between the adoption of an environmental management system with organisational environmental performance. *Australasian Journal of Environmental Management*, 21(3), 281-296.

- Tung, A., Baird, K., & Schoch, H. (2014b). The relationship between organisational factors and the effectiveness of environmental management. J Environ Manage, 144, 186-196.
- Tuzová, M., Toulová, M., & Kubíčková, L. (2017). The Specifics of the Internationalization Process of Czech SMEs in the Food Industry. Acta Universitatis Agriculturae et Silviculturae Mendelianae Brunensis, 65(3), 1055-1064.
- Umrani, W. A., Kura, K. M., & Ahmed, U. (2018). Corporate entrepreneurship and business performance: The moderating role of organizational culture in selected banks in Pakistan. *PSU Research Review*, 2(1), 59-80.
- UN. (2015). Transforming Our World: the 2030 Agenda for Sustainable Development. Retrieved 9-9-2018, from https://sustainabledevelopment.un.org/post2015/transformingourworld/public ation
- UNIDO. (2017). Pollution from food processing factories and environmental protection. Retrieved 15-02-2017, from http://www.unido.org/fileadmin/import/32129_25PollutionfromFoodProcessing http://www.unido.org/fileadmin/import/32129_25PollutionfromFoodProcessing
- USAID. (2007). Draft Policy Analysis on the Competitive Advantage of the Food Processing Sector in Pakistan: Focus on Quality, Safety and Standards. Retrieved 10-12-2018
- Uzkurt, C., Kumar, R., Semih Kimzan, H., & Eminoğlu, G. (2013). Role of innovation in the relationship between organizational culture and firm performance: A study of the banking sector in Turkey. *European Journal of Innovation Management*, 16(1), 92-117.
- Valmohammadi, C. (2011). The impact of TQM implementation on the organizational performance of Iranian manufacturing SMEs. *The TQM Journal*, 23(5), 496-509.
- van Kemenade, E., & Hardjono, T. W. (2019). Twenty-first century Total Quality Management: the Emergence Paradigm. *The TQM Journal*, *31*(2), 150-166.
- Vílchez, V. F., & Darnall, N. (2016). Two are Better Than One: The Link Between Management Systems and Business Performance. *Business Strategy and the Environment, 25*(4), 221-240.

- Vinzi, V. E., Chin, W. W., Henseler, J., & Wang, H. E. (2010). Handbook of partial least squares: Concepts, methods and applications: Springer Science and Business Media.
- Wahga, A. I., Blundel, R., & Schaefer, A. (2018). Understanding the drivers of sustainable entrepreneurial practices in Pakistan's leather industry: A multilevel approach. *International Journal of Entrepreneurial Behavior & Research*, 24(2), 382-407.
- Wali, A. A., Deshmukh, S. G., & Gupta, A. D. (2003). Critical success factors of TQM: A select study of Indian organizations. *Production Planning & Control*, 14(1), 3-14.
- Wang, C.-H., Chen, K.-Y., & Chen, S.-C. (2012). Total quality management, market orientation and hotel performance: The moderating effects of external environmental factors. *International Journal of Hospitality Management*, 31(1), 119-129.
- Weckenmann, A., Akkasoglu, G., & Werner, T. (2015). Quality management history and trends *The TQM Journal*, *27*(3), 281-293.
- Wee, Y. S., & Quazi, H. A. (2005). Development and validation of critical factors of environmental management. *Industrial Management & Data Systems*, 105(1), 96-114.
- Westney, D. E., & Piekkari, R. (2019). Reversing the translation flow:moving organizational practices from Japan to the U.S. *Journal of Management Studies*, 0(ja).
- Wetzels, M., Odekerken-Schröder, G., & Van Oppen, C. (2009). Using PLS path modeling for assessing hierarchical construct models: Guidelines and empirical illustration. *MIS Quarterly*, 177-195.
- Wiengarten, F., & Pagell, M. (2012). The importance of quality management for the success of environmental management initiatives. *International Journal of Production Economics*, 140(1), 407-415.
- Winder, C. (2000). Integrating OHS, environmental, and quality management standards. *Qual Assur, 8*(2), 105-135.
- Wong-MingJi, D. J., Kessler, E. H., Khilji, S. E., & Gopalakrishnan, S. (2014). Cross-cultural comparison of cultural mythologies and leadership patterns *South Asian Journal of Global Business Research*, 31(1), 79-101.

- Wong, K. K. (2013). Partial Least Squares Structural Equation Modelling (PLS-SEM) Techniques Using SmartPLS. *Marketing bulletin*, 24(1-32).
- World Bank. (2006). Pakistan Strategic Country Environmental Assessment: World Bank.
- Wronka, M. (2013). Analyzing the Success of Social Enterprises: Critical Success Factors Perspective. In Active Citizenship by Knowledge Management & Innovation : Proceedings of the Management, Knowledge and Learning International conference pp. 593–605. Toknow Press.
- WTO. (2016). An introduction to trade and environment in the WTO. Retrieved 06-12-2016, from

https://www.wto.org/english/tratop_e/envir_e/envt_intro_e.htm

- Wu, S. J. (2015). The impact of quality culture on quality management practices and performance in Chinese manufacturing firms. *International Journal of Quality & Reliability Management*, 32(8), 799-814.
- Wulandari, R. S., Soemirat, J., & Rahardyan, B. (2012). Analysis of Success Factors of ISO 14001 Implementation – A Case Study in a Cement Company in Indonesia. Paper presented at the Southeast Asian Network of Ergonomics Societies Conference (SEANES), Indonesia.
- Yeap, J. A. L., Ramayah, T., & Soto-Acosta, P. (2016). Factors propelling the adoption of m-learning among students in higher education. *Electronic Markets*, 26(4), 323-338.
- Yeng, S. K., Jusoh, M. S., & Ishak, N. A. (2018). The impact of total quality management (TQM) on competitive advantage: a conceptual mixed method study in the Malaysia luxury hotel industries. *Academy of Strategic Management Journal*, 17(2), 1-9.
- Yoo, D. K., Rao, S. S., & Hong, P. (2006). A comparative study on cultural differences and quality practices – Korea, USA, Mexico, and Taiwan. *International Journal of Quality & Reliability Management, 23*(6), 607-624.
- Yuan, L., Nguyen, T.-T.-N., & Vu, M.-C. (2018). Transformational leadership and its impact on performance. Association for Computing Machinery, 18-27.
- Yunis, M., Jung, J., & Chen, S. (2013). TQM, strategy, and performance: a firmlevel analysis. *International Journal of Quality & Reliability Management*, 30(6), 690-714.

- Yusof, S. R. M., & Aspinwall, E. M. (2000). Critical success factors in small and medium enterprises: Survey results. *Total Quality Management*, 11(4-6), 448-462.
- Zairi, M. (2013). The TQM legacy Gurus' contributions and theoretical impact. *The TQM Journal*, 25(6), 659-676.
- Zakuan, N. (2009). "Structural analysis of total quality management, ISO/TS 16949 and organizational performance in Malaysian and Thailand automotive industry", PhD thesis,. Universiti Teknologi Malaysia, Johor Bahru.
- Zakuan, N. M., Yusof, S. M., Laosirihongthong, T., & Shaharoun, A. M. (2010). Proposed relationship of TQM and organisational performance using structured equation modelling. *Total Quality Management & Business Excellence*, 21(2), 185-203.
- Zhang, H., & Yang, F. (2016a). On the drivers and performance outcomes of green practices adoption. *Industrial Management & Data Systems*, 116(9), 2011-2034.
- Zhang, H., & Yang, F. (2016b). On the drivers and performance outcomes of green practices adoption: An empirical study in China. *Industrial Management & Data Systems*, 116(9), 2011-2034.
- Zhang, X., & Vesselinov, V. V. (2017). Integrated modeling approach for optimal management of water, energy and food security nexus. *Advances in Water Resources, 101*, 1-10.
- Zhang, Z., Waszink, A., & Wijngaard, J. (2000). An instrument for measuring TQM implementation for Chinese manufacturing companies. *International Journal* of Quality & Reliability Management, 17(7), 730-755.
- Zhu, Q., Sarkis, J., & Geng, Y. (2005). Green supply chain management in China: pressures, practices and performance. *International Journal of Operations & Production Management*, 25(5), 449-468.
- Zubair, S. S. (2013). Total Quality Management in Public Sector Higher Education Institutions. *Journal of Business & Economics*, 5(1), 24-55.
- Zutshi, A., & Sohal, A. S. (2004). Adoption and maintenance of environmental management systems Critical success factors. *Management of Environmental Quality: An International Journal, 15*(4), 399-419.

Zutshi, A., & Sohal, A. S. (2005). Integrated management system: The experiences of three Australian organisations. *Journal of Manufacturing Technology Management, 16*(2), 211-232.

Appendix A Letter to SMEDA



Appendix B Letter to PSQCA



Appendix C Questionnaire



Dear respondent!

I am conducting a research on "Quality and Environmetal Management (QEM) Practices among Small and Medium-Sized Food Processing Enterprises in Pakistan" and have designed this questionnaire to collect data from food processing SMEs of Pakistan.

Survey Objective: The main purpose of this research is to investigate the Quality and Environmental Management (QEM) practices among SMEs of food processing in Pakistan. The responses from the questionnaire will be used to analyze the impact of QEM practices on operational and environmental performances of SMEs. The results will help in improving performance of SMEs. The outcome of the study will also help SMEs for future sustainable development.

Respondents Individuals: Owner, CEO, Manager Quality Assurance, Manager Operation, Manager Production, Senior Engineer or a most relevant person is requested to complete this questionnaire.

Data Confidentiality: The provided information will remain confidential and only be used for research purpose. Your contribution to complete the questionnaire is highly important for this study and also for the improvement of SMEs performance. Therefore, your cooperation and participation is highly appreciated.

In case of any query and concern please do not hesitate to contact the below mentioned addresses.

Sincerely,

Syed Ali Raza Shah PhD Candidate, UTM Razak School of Engineering and Advanced Technology, Universiti Technologi Malaysia International Campus, Jalan Semarak, 54100 Kuala Lumpur. E-mail: <u>razadopasi@gmail.com</u> Alternative e-mail: <u>arssyed2@live.utm.my</u> Cell Phone: +60124017554 or +923313938350

Associate Professor: Dr. Khairur Rijal Jamaludin Deputy Dean (Academic and student development) / PhD Supervisor. Khairur.kl@utm.my

SECTION 1: DEMOGRAPHICS INFORMATION

Please tick the most appropriate answer

(A). About Respondent

1. Education Background / qualification:

□ Higher Secondary	\square Bachelor's	□ Master's	\Box other qualification (pl.
specify)			

2. Experience in this SME:

 \Box 1-5 years \Box 6-10 years \Box 11-20 years \Box More than 20 years

3. Current working department /unit in this SME:

 \Box Quality \Box Engineering \Box Production \Box other department (pl. specify)

(B). About Small and Medium Enterprise (SME)

4. Type of Food Product Manufacture/engagement:

□ Food processing□ Beverage processing□ Both (Food and Beverages) processing

5. Year of Establishment/age:

 \Box 1-5 years \Box 6-10 years \Box 11-20 years \Box More than 20 years

6. Number of permanent employees:

□ 1-9 □ 10-49 □ 50-250

7. Nature of SME ownership:

□ Partnership	□ Sole Proprietorship	□ Public Limited	□ Private Limited
---------------	-----------------------	------------------	-------------------

8. SME location (District and Province):

9. SME exporting any food items?

 \Box Yes

Exporting countries:

Exporting items and applied standards (if any):

 \Box No

SECTION 2: CRITICAL SUCCESS FACTORS OF QUALITY AND **ENVIRONMENTAL MANAGEMENT (QEM)**

Please indicate the degree of agreement on QEM practices by marking most suitable number based on the scale of 1 to 5 for each statement below. Kindly answer ALL questions.

A. LEADERSHIP (Top management commitment, efforts and participation towards quality and environmental improvement)						
LS1	Top management creates and communicates a vision focused on quality improvement.	1	2	3	4	5
LS2	Top management pursues long-term business success.	1	2	3	4	5
LS3	Top management anticipates change and makes plans to accommodate it.	1	2	3	4	5
LS4	Top management provides rewards, bonuses to staff on best performance.	1	2	3	4	5
LS5	Top management share similar beliefs about the future direction of the company.	1	2	3	4	5
LS6	Top management encourages employee involvement in QEM (quality and environmental management) improvement activities.	1	2	3	4	5
LS7	Top management provides the necessary resources to carry out activities efficiently.	1	2	3	4	5
LS8	Top management emphasizes the importance of customers.	1	2	3	4	5
B. EMPLOYEE MANAGEMENT (Employee involvement in decision making process of quality and environmental planning, provision of training, awareness and rewards)						
EM1	The company has enough financial and technical resources for employees training.	1	2	3	4	5
EM2	The company provides trainings to employees for improving their competencies.	1	2	3	4	5
EM3	Employees in the company are dedicated to their jobs.	1	2	3	4	5
EM4	The company empowers employees and takes employees satisfaction in consideration.	1	2	3	4	5
EM5	The company encourages employees to find and fix the problems and issues.	1	2	3	4	5
EM6	The company promotes teamwork.	1	2	3	4	5
EM7	The company encourages employees to update their knowledge and skills.	1	2	3	4	5
EM8	The company provides a safe and healthy work environment.	1	2	3	4	5
C. STRATEGIC PLANNING (Quality and environmental improvement planning, by addressing problem/issues, achieving current and future goals of the company)						
SP1	The company involves employees in QEM (quality and environmental management) planning.	1	2	3	4	5
SP2	Results are evaluated with plans, in order to make improvements.	1	2	3	4	5

SP3	The company has a clear strategic objectives and plans.		2	3	4	5
SP4	The company developed long-term and short-term plans and corresponding actions.		2	3	4	5
SP5	Every employee is committed to achieve company strategic objectives and plans	1	2	3	4	5
SP6	The company plans focused on adopting 'Best QEM Practices'.	1	2	3	4	5
SP7	The company always incorporates stakeholder's (customers/suppliers) requirements.	1	2	3	4	5
SP8	The continuous quality improvement is considered in the company strategy.	1	2	3	4	5
D. 1	NFORMATION MANAGEMENT (Availability of quality and QEM data for improvement of processes and pr	enviro oducts	nment)	tal dat	a, usii	ng
IM1	The company shares QEM (quality and environmental management) information to employees.	1	2	3	4	5
IM2	The company analyses its operational activities data for performance improvement.	1	2	3	4	5
IM3	The company uses QEM information to improve its key processes and products.	1	2	3	4	5
IM4	The company regularly measures QEM practice performance.	1	2	3	4	5
IM5	The company uses different procedures to ensure the reliability of data collection.		2	3	4	5
IM6	The company assesses the performance of departments and employees.	1	2	3	4	5
IM7	The information obtained in the company is used to improve QEM practices.	1	2	3	4	5
E. PR	COCESS MANAGEMENT (Ensuring quality in production pro problems and using quality and environmental information for	cess, m · proces	anage ss imp	ement rovem	of qui ient)	ality
PM1	The company regularly does a quality control work.	1	2	3	4	5
PM2	The quality is measured during all processes.	1	2	3	4	5
PM3	Quality is important in the production process.	1	2	3	4	5
PM4	The company put efforts to prevent errors during the phase of process planning.	1	2	3	4	5
PM5	The company regularly monitors the quality of products and processes.	1	2	3	4	5
PM6	The company always does process improvement.	1	2	3	4	5
PM7	The company takes corrective action immediately when there is a quality problem.	1	2	3	4	5
<i>F. S</i>	UPPLIER MANAGEMENT (Quality and environmental award high quality suppliers, good communication and collaborat	eness to tion wit	o supp th sup	liers, pliers,	selecti)	ing
SM1	Quality is the more important criteria in selecting suppliers.	1	2	3	4	5
SM2	The company assists suppliers in improving quality.	1	2	3	4	5
SM3	The company relies on high quality suppliers.	1	2	3	4	5

SM4	The company collaborates with suppliers to improve product quality and ensure on-time delivery.	1	2	3	4	5
SM5	The company has relations with suppliers on QEM (quality and environmental management) practices.	1	2	3	4	5
SM6	The company gives a feedback on the supplier performance.	1	2	3	4	5
SM7	The company treats the suppliers as partners.	1	2	3	4	5

G. CUSTOMER FOCUS (Fulfil customer needs and expectations, maintain good relationship, consider customer complaints and satisfaction on top priority)

CF1	Customer's feedback is used to improve company performance.	1	2	3	4	5
CF2	The company always maintains a close relationship with its customers.	1	2	3	4	5
CF3	The company actively seeks customer needs for product improvement activities.	1	2	3	4	5
CF4	Customer needs and expectations are effectively disseminated and understood by all staff members.	1	2	3	4	5
CF5	The company uses customer complaints as an input for process improvement.	1	2	3	4	5
CF6	The company collects complaints information from its customers.	1	2	3	4	5
CF7	Quality-related customer complaints are treated with top priority in the company.	1	2	3	4	5
CF8	Customer satisfaction is considered as a major responsibility of the company.	1	2	3	4	5

SECTION 3: ORGANIZATIONAL CULTURE

Please indicate the degree of agreement on Organizational culture by marking most suitable number.

(1= Strongly Disagree; 2 = Disagree; 3= Neutral; 4= Agree; 5= Strongly Agree).

OR	ORGANIZATIONAL CULTURE (Mutual respect and sharing of information among team members, willingness to accept change, team work)						
OC1	There is a mutual respect among team members.	1	2	3	4	5	
OC2	There is sharing of information among team members.	1	2	3	4	5	
OC3	There is willingness to accept change in the organizational structure.	1	2	3	4	5	
OC4	There is willingness to deal with customer queries in time.	1	2	3	4	5	
OC5	The company involve partners (suppliers and customers) in the decision making process.	1	2	3	4	5	
OC6	There is a participation and open discussion among all staff.	1	2	3	4	5	
OC7	The company considers employee ideas and concerns.	1	2	3	4	5	
OC8	There is a team work and team spirit among all staff.	1	2	3	4	5	

SECTION 4: ORGANIZATIONAL PERFORMANCE

Please indicate the degree of agreement on operational and environmental performance by marking most suitable number.

(1= Strongly Disagree; 2 = Disagree; 3= Neutral; 4= Agree; 5= Strongly Agree)

A. OPERATIONAL PERFORMANCE (Reduction in operational costs, increase in product							
	quality, on time delivery and increase in company image in the market)						
OP1	The company's operational cost is reducing.	1	2	3	4	5	
OP2	The company's products quality is increasing.	1	2	3	4	5	
OP3	The company's production quantity is increasing.	1	2	3	4	5	
OP4	The company's process time is reducing.	1	2	3	4	5	
OP5	The company's operational performance efficiency is increasing.	1	2	3	4	5	
OP6	The company's process effectiveness is increasing.	1	2	3	4	5	
OP7	The company's positive image is increasing.	1	2	3	4	5	
OP8	The company's product stock is reducing due to higher product selling.	1	2	3	4	5	
B. EN reduci	B. ENVIRONMENTAL PERFORMANCE (Improvement in environmental performance by reducing environmental pollution, solid waste, water wastage, less energy consumption)						
EP1	Environmental performance of the company is improving.	1	2	3	4	5	
EP2	Environmental management practices (EMS) help to improves company's image.	1	2	3	4	5	
EP3	The wastage of food items are reducing in the company.	1	2	3	4	5	
EP4	The amount of water wastage in different processes is reducing in the company.	1	2	3	4	5	
EP5	The consumption of the harmful material is decreasing in the company.	1	2	3	4	5	
EP6	The environmental pollution is reducing in the company.	1	2	3	4	5	
EP7	The consumption of water in the company is reducing.	1	2	3	4	5	
EP8	The consumption of energy in the company is decreasing.	1	2	3	4	5	

-END OF QUESTIONNAIRE-

Appendix D Data Collection Letter

6	UTM Razak School of Engineering and Advanced Techr UTM Kuala Lumpur	ology	UTM Razak School of Engineering and Advanced Technology Level 7, Menara Razak Universiti Teknologi Malaysia Jalan Sultan Yahya Petra 54100 Kuala Lumpur, Malaysia
	Tel: +(6)03-21805138 Fax: +(6)0	3-21805380 http:/www.r	azakschool.utm.my
OUR REF.:	UTM.K.40.02 /13.1	1/1/4/ Jld. 8 (9 3)	7 November 2017
	TO WHOM IT MAY CONCERN		
	Dear Sir/ Madam,		
	DATA COLLECTION FOR RESEARCH		
	This is to certify that the bearer of the Engineering and Advanced Technology who is currently pursuing Doctor of Phicould assist his field work in order to example the SYED ALI RAZA	his letter is a student a gy, Universiti Teknologi N ilosophy. It would be gre eecute his research at yo SHAH	t UTM Razak school of Aalaysia, Kuala Lumpur, aatly appreciated if you ur company.
	ISID No : 201604M10008		
	Matric No : PR\$153048		
	Project title : QUALITY AND E PRACTICES AMO PROCESSING EN	NVIRONMENTAL MANAG DNG SMALL AND MEDIUN ITERPRISES IN PAKISTAN	EMENT (QEM) I-SIZED FOOD
	Supervisor : ASSOC. PROF. I	R. KHAIRUR RIJAL BIN JA	MALUDIN
	Supervisor's : khairur.kl@utm. Email	my	
	Supervisor's : +60321805130 / Contact No.	+60193862305	
	Please do not hesitate to contact his	supervisor for further infor	mation.
	Thank you for your kind support.		
	Yours sincerely, NASIR BIN OSMAN Deputy Registrar UTM Razal School of Engineering and UTM Kuala Lumpur For The Vice Chancellor © 03-21805360 03-21805380 Nasir kl@utm.my	Advanced Technology	
	cc. Supervisor		
	NUTRICUL AND	MEDICIA ACADO 2001 Rending Science & Factories Science & Factories	
	www.utm.r	ny innovative • er	itrepreneurial • global

Appendix E Pilot Study Letter

6	UTTM NIVERSITI TEKNOLOGI MALAYSIA	M Razak School of gineering and Advanced Technology M Kuala Lumpur	UTM Razak School of Engineering and Advanced Technology Level 7, Menara Razak Universiti Teknologi Malaysia Jalan Sultan Yahya Petra 54100 Kuala Lumpur, Malaysia
	Tel: +(6)03-218	305138 Fax: +(6)03-21805380 http://www	v.razakschool.utm.my
OUR REF.:	UTM.K.40.02	.03/13.11/1/4 Jld. 7 (98)	29 May 2017
	TO WHOM IT MAY	CONCERN	
	Dear Sir/Madam,		
	DOCTOR OF PHIL	OSOPHY PROJECT	
	This is to certify Engineering and A who is currently p could assist him da	that the bearer of this letter is a stu Advanced Technology, Universiti Tekn- ursuing Doctor of Philosophy. It wou ata collection in your company.	dent at UTM RAZAK School of ologi Malaysia, Kuala Lumpur, ld be greatly appreciated if you
	Name :	SYED ALI RAZA SHAH	
	NRIC No. :	201604M10008	
	Matric No :	PRS153048	
	Project Title :	QUALITY AND ENVIRONMENTAL M AMONG SMALL AND MEDIUM FOO ENTERPRISES IN PAKISTAN	IANAGEMENT PRACTICES D PROCESSING
	Supervisor :	ASSOC PROF. DR. KHAIRUR RIJAI	BIN JAMALUDIN
	E-mail :	khairur.kl@utm.my	
	Telephone No:	03-2180 5130	
•	Please do not hesita	te to contact the supervisor above for fu	urther information.
	Thank you for your	kind support.	
	Yours sincerely,		
	Muni		
	NASIR BIN OSMAN Deputy Registrar UTM RAZAK School UTM Kuala Lumpur For The Vice Chance 03-2180 5360 03-2180 5360 03-2180 5860 <u>nasir.kl@utm.r</u>	of Engineering and Advanced Technolog Illor <u>ny</u>	ΣΥ
	c.c. Student file	EXCEPTION OF THE STATE OF THE S	2 300 W
		www.utm.my innovative •	entrepreneurial • global

Appendix F Chamber of Commerce Letter

THE MULTAN CHAMBER OF COMMERCE & INDUSTRY (RECOGNISED BY THE GOVERNMENT OF PAKISTAN)

Phone : 061-4517087 - 4543530 Fax : 061-4570463 E-mail : mccimultan@gmail.com Website : www.mcci.org.pk



Shahrah-e-Aiwan-e-Tijarat-o-Sanat Multan. (Pakistan)

Ref. No: MCCI/2017/0054.

Date: 25-07-2017.

TO WHOM IT MAY CONCERN

This is to certify that Mr. Syed Ali Raza Shah is pursuing his PhD in Quality Managementat UTM Razak School of Engineering and Advanced Technology, Universiti. Technologi Malaysia. His research objective is to investigate the quality and environmental management practices among food processing SME's in Pakistan. All the information provided will be confidential and will be used for research purpose only. Your cooperation, honest opinions and contributions to complete the questionnaire is very much appreciated.

Sincerely,

Khurram Javed Secretary General

SECRETARY GENERAL Multan Chamber of Commerce & Industry MULTAN

	Name/Position/Affiliation	Comments
1.	Dr. Muslim Amin	Section 2:
	Associate Professor Department of Management College of Business Administration King Saud University Kingdom of Saudi Arabia	 Part (A) Q1; it might be SMEs communicates vision on quality but not environmental otherwise. Part (B) Q1; you need to explain the resources means, please specify. Part (C) Q2; please make it clear, what is results? Part (D) Q7; Please make it clear about adjust policy. Part (E); Q1 and Q8 are redundant. Part (G); Q5, Q6, Q7 will make the respondent confuse. Please rewrite the sentence. Section 3:
2	Dr. Mahd Miarm Ah Dahman	• Q1 looks the same with Q8.
2.	Associate Professor Department of Mechanical and Materials Engineering Faculty of Engineering & Built Environment Universiti Kebangsaan Malaysia	 Covering letter to respondent s on page 1: Replace the sentence "The objective of this research" with the sentence "This research objective" and mention "investigate about". Section 2:
		 Part (A) Q3; this writing is not so clear. Please improve the statement. Part (A) Q8; this writing is not so clear. Please improve the statement. Part (B) Format; Font (Default) Times New Roman, 12 pt. Part (B) Q1; Need to mention "there are enough resources". Part (B) Q2; this is not so clear. Please improve the statement. Part (B) Q3; not clear. Please elaborate. Part (B) Q4; good one but please rewrite statement. Part (B) Q6; please rewrite and

Appendix G Summary of Experts Opinion

		clarify statement.
		• Part (B) Q8; good one.
		• Part (C) Q2; please rewrite with
		more clarity
		• Part (C); Q3 and Q4 should be
		will
		• Part (C) O6: please rewrite
		 Part (D) Q1: Please check and try
		to use another appropriate word.
		• Part (D) Q4; please rewrite more
		clearly.
		• Part (E) Q3; please rewrite more
		clearly.
		• Part (E) Q4; please rewrite more
		clearly.
		• Part (E) Q8; please rewrite more
		• Part (F) O1: please rewrite more
		clearly.
		• Part (G) Q2; please rewrite more
		clearly.
		Section 3:
		• O1: please double check regarding
		clarity.
		Section 4:
		• Part (B) Q1; please rewrite with
		more clarity.
		• Part (B) Q6; please rewrite with
		more clarity.
		• Part (B) Q8; please double check
		uns. Last para needs double check for
		alamiter
		ciarity.
3.	Dr. Muhammad Noman Malik	Section 1:
	Assistant Professor	• Q5; mention province and
	Department of Computer Science	district/cities. Section 2:
	Faculty of Engineering & Computer	• Dort (C) O1: Donbross, not alors
	Science	• Fait (C) Q1; Rephrase, not clear as users will get confusion while
	National University of Modern	regarding this.
	Languages	and understanding issue
	Islamahad Campus Dakistan	• Part (C) O5: not meaningful.
	isiainavau Vainpus, rakistan	• Part (C) Q8; item 2 and 8 seems
		same objective.

		 Part (D) Q2; mention " its operational activities" Part (D) Q4; needs to rephrase for more clarity. Part (D) Q5; item 5 and 6 has contrary answers. Rephrase them. Part (D) Q8; this should be in performance measurement. Part (E) Q8; similar objective as item 5 and 6. Part (F) Q4; item 4 and 2 has same objectives Rephrase them. Part (G) Q3; item 2 and 3 has same meaning. Rephrase them Q8; not meaningful. Rephrase it. Section 4:
		 Part (A) Q1; in cost of what? Not clear. Part (A) Q5; in term of what? Not meaningful. Rephrase it. Part (A) Q8; not meaningful. Part (B) Q7; need to remove or rephrase. Part (B) Q8; not meaningful.
4.	Professor Dr. Nawar Khan Senior Consultant PhD Engineering Management ISO 9000: 2015 Lead Assessor Faculty of Management Sciences Riphah International University Al-Mizan Campus, Rawalpindi Pakistan	 Section 1: Part (A); mention "Please tick the most appropriate answer" Part (A) Q2; mention the word "since". Part (A); Add a question related to "export involvement" Part (B) Q4; replace the word "organization" with "SME". Section 2: Correct the grammar of content. Part (B) Q2; instead of using "our company" use "the company" Part (B) Q4, Q5, Q6 replace "our company" with "this company" Part (C) Q1, Q2; rephrase, correct the words for more clarity. Part (C); Q3, Q4, Q5, Q6, Q7 and Q8 change the word "our

		 Part (E); replace the word "our company" with "the company" in all questions. Part (F); replace the word "our company" with "the company" in all questions. Part (G); replace the word "our company" with "the company" in all questions. Section 4: Correct the grammar from "our company" to "this company". Part (A) Q3; replace the word" delivery" with quantity. Mention the sentence in last para. "Your opinion will be kept secret and mixed with other data to avoid individual bias and identity".
5.	Rana Moazam Maqbool	Section 2:
	Project-Quality Assurance Manager Dutch Mill (Pyt) Ltd	• Part (C); Q2 is not clear. Section 3:
	Jhang Pakistan	• Q1 and Q8 have same objectives. Section 4:
		• Part (A) Q4; needs to be more clear.
6.	Awais Ali Manager Quality	• I have gone through the questionnaire you shared, it seems well to me. So I am agreed on it
	Gourmet Foods Lahore Pakistan	for further proceedings.
	Lunor, i unioun	

	Zscore						
	Ν	Minimum	Maximum				
Zscore(LS1)	288	-2.95834	1.34470				
Zscore(LS2)	288	-3.04331	1.31725				
Zscore(LS3)	288	-2.79030	1.32548				
Zscore(LS4)	288	-2.80055	1.29366				
Zscore(LS5)	288	-2.86902	1.44575				
Zscore(LS6)	288	-3.11867	1.43482				
Zscore(LS7)	288	-2.91646	1.39093				
Zscore(LS8)	288	-2.91172	1.37219				
Zscore(EM1)	288	-3.03522	1.18769				
Zscore(EM2)	288	-2.87163	1.25835				
Zscore(EM3)	288	-3.43816	1.23806				
Zscore(EM4)	288	-3.35816	1.12978				
Zscore(EM5)	288	-3.13981	1.26587				
Zscore(EM6)	288	-3.41080	1.23369				
Zscore(EM7)	288	-3.21035	1.21878				
Zscore(EM8)	288	-3.06494	1.26734				
Zscore(SP1)	288	-2.84236	1.31297				
Zscore(SP2)	288	-2.75480	1.25218				
Zscore(SP3)	288	-2.88980	1.35103				
Zscore(SP4)	288	-2.82908	1.42007				
Zscore(SP5)	288	-2.80145	1.49585				
Zscore(SP6)	288	-2.59190	1.34205				
Zscore(SP7)	288	-2.86916	1.39021				
Zscore(SP8)	288	-3.19076	1.29734				
Zscore(IM1)	288	-2.92010	1.31129				
Zscore(IM2)	288	-2.85643	1.33543				
Zscore(IM3)	288	-2.69392	1.30530				
Zscore(IM4)	288	-2.58734	1.28862				
Zscore(IM5)	288	-2.54008	1.40887				
Zscore(IM6)	288	-2.93406	1.40488				
Zscore(IM7)	288	-2.73105	1.22142				
Zscore(PM1)	288	-2.79488	1.33294				
Zscore(PM2)	288	-3.03697	1.26634				
Zscore(PM3)	288	-2.73933	1.27556				
Zscore(PM4)	288	-2.95029	1.35736				
Zscore(PM5)	288	-2.90870	1.55312				
Zscore(PM6)	288	-2.92208	1.27522				
Zscore(PM7)	288	-2.93785	1.28737				
Zscore(SM1)	288	-2.50436	1.30174				

Appendix H Z Scores

Zscore(SM2)	288	-2.79673	1.38204
Zscore(SM3)	288	-2.80227	1.37935
Zscore(SM4)	288	-2.74408	1.37741
Zscore(SM5)	288	-2.73909	1.42375
Zscore(SM6)	288	-2.68071	1.34036
Zscore(SM7)	288	-2.84015	1.24456
Zscore(CF1)	288	-2.95993	1.33458
Zscore(CF2)	288	-3.10818	1.29062
Zscore(CF3)	288	-3.05729	1.35072
Zscore(CF4)	288	-3.02852	1.32167
Zscore(CF5)	288	-2.98600	1.25552
Zscore(CF6)	288	-3.05398	1.27876
Zscore(CF7)	288	-3.18210	1.29929
Zscore(CF8)	288	-3.07292	1.17169
Zscore(OC1)	288	-3.13857	1.21761
Zscore(OC2)	288	-3.07511	1.22407
Zscore(OC3)	288	-3.13555	1.35718
Zscore(OC4)	288	-3.03924	1.24096
Zscore(OC5)	288	-2.99776	1.26047
Zscore(OC6)	288	-3.09446	1.31199
Zscore(OC7)	288	-3.05083	1.23000
Zscore(OC8)	288	-2.91035	1.13395
Zscore(OP1)	288	-2.81974	1.23562
Zscore(OP2)	288	-3.19177	1.23825
Zscore(OP3)	288	-3.05398	1.27876
Zscore(OP4)	288	-2.98054	1.32759
Zscore(OP5)	288	-2.83525	1.28875
Zscore(OP6)	288	-3.34720	1.36670
Zscore(OP7)	288	-2.46131	1.17385
Zscore(OP8)	288	-2.93592	1.18366
Zscore(EP1)	288	-2.71893	1.13847
Zscore(EP2)	288	-2.91963	1.11808
Zscore(EP3)	288	-3.07738	2.03827
Zscore(EP4)	288	-2.79975	1.22686
Zscore(EP5)	288	-2.75566	1.13946
Zscore(EP6)	288	-2.84585	1.15222
Zscore(EP7)	288	-2.86102	1.25371
Zscore(EP8)	288	-2.95776	1.20769
Valid N (listwise)	288		

LIST OF PUBLICATIONS

Journals:

- Syed Ali Raza Shah, Jamaludin K.R, Hayati Habibah Abdul Talib, and Sha'ri Mohd Yusof. Integrated Quality Environmental Management Implementation in Food Processing SMEs: A Case Study of a Developing Country, *The TQM Journal* (Accepted for Publication) (Q2)
- Syed Ali Raza Shah, Jamaludin K.R, and Hayati Habibah Abdul Talib. Assessing Environmental Management Practices in Food Processing SMEs of Pakistan by using Partial Least Squares (PLS). *British Food Journal* (in review)(Q1)
- Syed Ali Raza Shah, Jamaludin K.R, and Hayati Habibah Abdul Talib. Quality Management Implementation and Their Impact on Operational Performance of Pakistan's Food Processing SMEs. *International Journal for Quality Research* (Accepted with minor revision, revised and resubmitted).(Scopus)

Conference Proceeding:

 Jamaludin K.R, Hayati Habibah Abdul Talib and Syed Ali Raza Shah. Integrated Quality Environmental Management Practices and its Impact on operational and environmental performance in Food Processing SMEs *International Conference on Industrial Engineering and Operations Management*, March 5-7, 2019 Bangkok, Thailand (Received Best Track Paper Award).(Scopus Indexed)

- Jamaludin K.R, Syed Ali Raza Shah, Hayati Habibah Abdul Talib, and Sha'ri Mohd Yusof Quality Management Practices among Small and Medium-Sized Food Processing Enterprises in Pakistan. *Asia Pacific Industrial Engineering* & *Management Systems Conference*, December 5-8, 2018 Hong Kong (Scopus Indexed)
- Syed Ali Raza Shah, Jamaludin K.R, and Hayati Habibah Abdul Talib. Preliminary Study on Environmental Management Practices among Small and Medium Food Processing Enterprises in Pakistan. Asia International Multidisciplinary Conference, May 1-2, 2017 at UTM Johor Bahru Campus.